Making a way when there is no way:

The experiences and challenges of gang affected young adult refugees in Winnipeg

by

Matthew Fast

A Thesis submitted to the Faculty of Graduate Studies of
The University of Manitoba

In partial fulfillment of the requirements of the degree of

MASTER OF ARTS

Department of Peace and Conflict Studies

University of Manitoba

Winnipeg

Copyright © 2013 by Matthew Fast

THE UNIVERSITY OF MANITOBA

FACULTY OF GRADUATE STUDIES

COPYRIGHT PERMISSION

Making a way when there is no way: The experiences and challenges of gang affected young adult refugees in Winnipeg

By

Matthew Fast

A Thesis submitted to the Faculty of Graduate Studies of University of Manitoba in partial fulfillment of the requirement of the degree Of Master of Arts in Peace & Conflict Studies (M.A.)

(c) 2013 by Matthew Fast

Permission has been granted to the Library of the University of Manitoba to lend or sell copies of this thesis, to the National Library of Canada to microfilm this thesis and to lend or sell copies of the film. and the University Microfilms Inc. to publish an abstract of this thesis/practicum.

This reproduction or copy of this thesis has been made available by authority of the copyright owner solely for the purposes of private study and research, and may only be reproduced and copied as permitted by copyright laws or with express written authorization from the copyright owner.

Abstract

This study explores the perceptions, and life experiences of formerly ganginvolved young adult refugee men living in Winnipeg. In doing so, this study examines both the negative experiences and challenges of these young men that led to their involvement with gangs, and the positive and life changing events that provided the catalyst for these young men to leave their gangs. If positive support mechanisms are insufficient, and if their basic human needs cannot be satisfied, refugee young people become at-risk of involvement in antisocial behavior and criminal activity. In order to assist refugee young people in their successful transition into a foreign culture and society it is essential to understand how their perceptions and experiences inform their identity and behavior. This study contributes to this understanding, which will inform policy and future approaches by government and community-based organizations to assist them in their transition.

Acknowledgements

First, I would like to thank Dr. Sean Byrne for his guidance and encouragement throughout this thesis process. I would also like to thank Dr. Jessica Senehi and Dr. Judith Harris for their added insight and helpful suggestions along the way. I would like to thank Melanie Zurba for her words of wisdom, inspiration and encouragement. I would also like to express gratitude to the staff and my colleagues at the Mauro Centre. Thanks to my family for all of their years of support and encouragement. Last, but certainly not least I would like to thank my participants. This thesis would not have been possible without your willingness to share your experiences and stories with me. I cannot express enough my gratitude and upmost respect for all of you.

Table of Contents

| Abstract | i |
|---|----------|
| Acknowledgements | ii |
| Chapter 1 – Introduction | 1 |
| 1.1 ó Introduction | 1 |
| 1.2 ó Purpose Statement | 2 |
| 1.3 ó Significance of the Study | 2 |
| 1.4 ó Limitations of the Study | 3 |
| 1.5 ó Context of the Study | 5 |
| 1.5.1 ó Refugees in a Global Context | 5 |
| 1.5.2 ó Refugees in a Winnipeg Context | 6 |
| 1.5.3 ó Refugees in Winnipegøs Inner City 1.5.4 ó Challenges for Young Refugees in Winnipeg | 7 8 |
| 1.5.4 o Chantenges for Toung Refugees in Willington 1.6 o Framework of the Study | 10 |
| 1.7 ó Conclusion | 12 |
| | 12 |
| Chapter 2 – Literature Review | 13 |
| 2.1 ó Introduction | 13 |
| 2.2 ó Youth and Warfare | 13 |
| 2.3 ó Experiences with Direct Physical Violence | 16 |
| 2.4 ó Structural Violence | 17 |
| 2.5 ó Basic Human Needs | 19 |
| 2.5.1 ó School/Education | 22 |
| 2.5.2 ó Family | 24 |
| 2.5.3 ó Community 2.5.4 ó Peer/Social Group | 26 26 |
| 2.5.5 ó Identity/Behavior | 28 |
| 2.6 ó Conclusion | 30 |
| | |
| Chapter 3 – Methodology | 31 |
| 3.1 ó Introduction | 31 |
| 3.2 ó Qualitative Research Strategy | 31 |
| 3.3 ó Semi Structured Interviews | 33 |
| 3.4 ó Study Participants | 34 |
| 3.5 ó Location of the Study | 35 |
| 3.6 ó Role of the Researcher | 36 |
| 3.7 ó Data Gathering Techniques | 37 |
| 3.8 ó Data Analysis | 38 |
| 3.9 ó Conclusion | 43 |

| Chapter 4 – Down the Path to Gang Life | 45 |
|---|-----|
| 4.1 ó Introduction | 45 |
| 4.2 ó Adapting to School | 46 |
| 4.3 ó Perceptions of the Neighborhood | 52 |
| 4.4 ó Peer Group | 55 |
| 4.5 ó Challenges in the Family | 57 |
| 4.6 ó Perceptions in the Ethnic Community | 64 |
| 4.7 ó Perceptions of Law Enforcement | 68 |
| 4.8 ó Key Findings of the Research | 73 |
| 4.9 ó Conclusion | 79 |
| Chapter 5 – Walking Away from Gang Life | 81 |
| 5.1 ó Introduction | 81 |
| 5.2 ó Influences and Role Models | 81 |
| 5.3 ó Turning Points | 88 |
| 5.4 ó Coping Strategies | 91 |
| 5.5 ó Addressing the Issues | 94 |
| 5.6 ó Key Findings of the Research | 97 |
| 5.7 ó Conclusion | 100 |
| Chapter 6 – Finding a Way Forward | 101 |
| 6.1 ó Introduction | 101 |
| 6.2 ó Reflecting on Key Findings | 101 |
| 6.3 ó Future Research | 111 |
| 6.4 ó Final Conclusion | 113 |
| References | 115 |
| Appendix A | 122 |

Chapter One - Introduction

1.1 – Introduction

Ongoing protracted conflicts in various regions of the world have led to annual increases in the number of people living in refugee situations. Increasingly, Winnipeg, Canada is becoming home to many of these refugees. Refugees, and in particular refugee young people, are vulnerable as they face challenges integrating into their new environment (Kanu, 2008; Rossiter & Rossiter, 2009; Sersli, Salazar, & Lozano, 2010). As well, refugee young people are exposed to various forms and degrees of violence, that can have negative consequences for their personal development (Kaplan, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010). If positive support mechanisms are insufficient, and if their basic human needs cannot be satisfied, refugee young people become at-risk of involvement in antisocial behavior and criminal activity (Chekki, 2006; Kanu, 2008; Rossiter & Rossiter, 2009; Sersli et al., 2010).

This study explores the perceptions, and life experiences of war-affected refugee young adult males living in Winnipeg with regards to their challenges, struggles and coping strategies. In particular, the focus of this study is on four young adult refugee males who have previously fallen into a pattern of anti-social behavior where they have engaged in gang life and criminal activity, and who have now decided to make positive changes in their lives. This study also includes the perspectives of five community workers, three of whom are males, and two of whom are females. As well, this study also includes the voices of one community elder, and a Winnipeg Police Service officer.

1.2 – Purpose Statement

The purpose of this study is to provide a better understanding of the impact of the various life experiences and challenges on these refugee young people, which have ultimately contributed to their antisocial behavior and gang involvement. I also indentify some of the positive experiences, catalysts and motivations, which have initiated positive changes in their lives. Learning how these experiences and perceptions inform the identity and behavior of refugee young people is essential for community agencies, social workers, educators, mentors, and policymakers in providing the necessary inputs required for their successful transition into a foreign culture and society.

1.3 - Significance of the Study

As the number of refugees coming into Canada, and Winnipeg in particular continues to rise, research on refugee young people is becoming all the more important (Carter, Polevychok, & Friesen, 2006; Kanu, 2008; Neumann, 2011). This thesis highlights some of the significant and complex challenges young refugees face when integrating into a new society. When young people do not have the necessary support mechanisms, the challenges they face can put them at-risk of taking part in antisocial behavior and criminal activity.

It is my hope that this study will contribute to the growing research and literature about refugees and refugee young people. This contribution will add to the already existing theory on the relationship of access to basic human needs both physical needs such as food, clothing, and shelter, as well as social needs such as belonging, love and self-esteem and the root causes of anti-social behavior, and various forms of violence.

Furthermore, this study not only highlights the perceptions, experiences, and challenges of refugee young people, but it also identifies coping strategies, and turning points which reflect these young people's decision to make positive changes in their lives. This knowledge can inform government and non-profit policy with regards to funding and appropriate programming for these young people and their families.

Moreover, this study contributes to the literature on youth gangs, and in particular youth gangs in Winnipeg of which there is very little available literature. The recent gang related shooting death of an individual in Winnipeg on February 15th, 2013 in which a young Sudanese man was arrested (Turner, 2013), underscores the relevance, timing, and importance of this study. The knowledge gained from this study will also be useful in informing the development of any gang prevention or gang-exit programs for other areas with similar demographics and challenges.

Finally, and very importantly, this study provides refugee young people with an opportunity to have their voices heard and listened to. This process in and of itself can be considered an empowering and liberating experience as these young people have the space to tell their stories, and to feel heard by myself (the researcher) as well as by community workers, law enforcement officers, policymakers, and academics who will have the opportunity to read this study.

1.4 - Limitations of the Study

There are several limitations to this study. First, the young men who participated in this study were all of African descent. The study was not intentionally designed this way however, the key informants used in this study were more connected to the various

African communities within Winnipeg. This obviously leaves out the perspectives, experiences and challenges of other young adult refugees from places like Afghanistan and Myanmar (Burma).

Second, none of the participants in this study knew how to speak English or French prior to coming to Canada. French speaking refugees from Democratic Republic of Congo or Sierra Leone for example, may have a completely different experience settling into Winnipeg given Winnipegøs large French speaking community in St. Boniface, and the many French speaking schools that young people have the opportunity to attend. It would have been beneficial to note whether French speaking refugee young people have an easier time adapting and integrating into Winnipeg as opposed to their counterparts who only speak Arabic in addition to their mother tongue. Several attempts were made to locate appropriate participants with the help of a Francophone new-comer organization and a Francophone school however, these attempts were fruitless.

Third, the refugee young people who participated in this study were all males. The original design of this study called for female participants to be included as well. Finding female participants who not only fit with the criteria of the study, and who were also willing to share their stories with the researcher proved to be a difficult task. Two potential female participants were contacted and interview times were arranged with both of them. However, both of the young women failed to show up for their interviews. The interviews were rescheduled, and neither of the young women managed to show up. As a result, I decided to proceed with the study including only having male participants. The participation of females in the study would have provided a well-rounded and more informed thesis.

Fourth, although this thesis includes the perspectives of four young refugee men as well as five community workers, a police officer, and a community elder in order to provide a more layered analysis of some of the experiences and challenges of young refugees, it does not include the perspectives of school teachers or family members. Again, including school teachers and family members would have provided a fuller and more informed thesis.

1.5 – Context

1.5.1 – Refugees in a Global Context

I don't know if anyone really knows war until it lives inside of them. A person can come in and see the war, fear the war, be scared of the violence – but their life, their very being, is not determined by the war. This is my country, the country of my parents, my family, my friends, my future. And the war has gotten into all of these...for I can never leave the war. I will carry the war with me.

- A woman from Mozambique (Nordstrom, 1997, p. 7-8)

In 2010 the United Nations (UN) reported that there were 25 protracted conflict situations in 21 different countries (*United Nations 65th General Assembly Report*, 2010). The continued presence of protracted conflict situations has impacted people worldwide who are forced to flee their homes in greater numbers. By the end of 2010 the United Nations High Commission for Refugees (UNHCR) reported the second highest number of refugees on record, stating that there were 10.55 million refugees under its care, an increase of 153,000 refugees from 2009 (UNHCR, 2010a). Ongoing conflicts in sub-Saharan Africa, and in particular the continued volatile situation in Somalia account for most of the increase in refugee populations globally (UNHCR, 2010a).

Recently, the UNHCR declared that nearly half of the refugees under its care are children (UNHCR, 2010b), underscoring the importance of understanding the challenges

and vulnerability youth face within the refugee context. In many cases, refugee youthsø schooling is disrupted as a result of being displaced by war (Kanu, 2008; UNHCR, 2010b). As well, thousands of war-affected youth, most notably in the African context, are forced into becoming child soldiers or sex slaves (Beah, 2007; Kanu, 2008; Wessells, 2006; Wilkinson, 2002). The unimaginable atrocities either committed or incurred in these situations leave refugee youth with lingering memories and traumatic scars of their experiences (Kanu, 2008). These experiences create a fear and distrust among refugee youth of the police and other authority figures in society (Macdonald, 2007). The negative perception of authority figures among many refugee youth compounds the barriers and challenges they face upon arrival in a new host community. Winnipeg is increasingly becoming a host community for many young refugees who face these same challenges.

1.5.2 - Refugees in a Winnipeg Context

In previous decades the make-up of immigrants coming to Winnipeg reflected the largely European source countries. However, statistics have drastically changed over the past several years. Kanu (2008) states that a 2006 Manitoba Labour and Immigration study noted that in the last five years leading up to the study, seven of the top ten refugee home countries were in Africa. The past several years have also seen a general increase in immigration, including refugee intake in Manitoba. Between 1998-2005 the number of immigrants who came to Manitoba increased from 3,000 to 8,200 (Carter et al., 2006). Kanu (2008) again cites the 2006 Manitoba Labour and Immigration study, noting that in the same time period between 1998-2005 the number of refugees among the immigrant

population was 8,190, half of which were between 0-24 years-of-age (Kanu, 2008). This figure highlights the need to acknowledge and recognize the unique experiences, vulnerabilities, challenges and needs of young refugees. In 2010 the number of new immigrants to Manitoba reached 15,803 (Neumann, 2011), and 70-80 percent of these new immigrants choose to settle in Winnipeg (Carter et al., 2006; Neumann, 2011).

1.5.3 - Refugees in Winnipeg's Inner City

Winnipegos inner city has increasingly become home to many people in the refugee community (Carter, 2009a; Stewart, 2011). In 2009 the population of Winnipegos inner city was 120,000 making up 19 percent of the cityos population (Carter, 2009a). Within this same year, 24,000 people living in Winnipegos inner city were classified as ÷visible minoritiesø and 4,500 were identified as ÷recent immigrantsø having arrived within the last five years (Carter, 2009a). In addition, a 2006 study found that 22 percent of Winnipegos inner city population were foreign born, and 20 percent of its new arrivals were refugees (Carter et al., 2006).

Winnipegos inner city has faced an era of urban decay alongside pronounced marginalization and poverty for many who live there. A 2009 study indicated that 33 percent of families in the inner city live below the poverty line, well above the city average of 19 percent (Carter, 2009a). As well, a recent study on the resettlement experience of privately sponsored refugees showed that 68 percent of privately sponsored refugees lived in poverty, more than 3 times the level of average Winnipeg households (Carter, 2009b). The same study found that there has not been a significant increase in income with time spent in Winnipeg as poverty and inadequate employment levels

remain high among refugees no longer considered new arrivalsø (Carter, 2009b). Although many refugee households have cited concerns over their personal safety within the inner city, as well as their desire to move to safer neighborhoods, they often lack the financial resources and professional skills that would empower them to lift themselves out of poverty and move to a more desired neighborhood (Carter, 2009a).

The socioeconomic marginalization experienced by many refugee families living in Winnipegos inner city leaves refugee youth vulnerable to gang recruitment. A 2004 study carried out by *The Winnipeg Foundation* identified a correlation between the area where most gang activity occurs and the representation of ethnic minority youth in the area (Neumann, 2011). Moreover, a 2002 study on youth gangs within Canada suggests that 82 percent of the youth involved in gangs are visible minorities representing African, Asian, Hispanic and East Indian ethnic groups (Rossiter & Rossiter, 2009).

1.5.4 - Challenges for Young Refugees in Winnipeg

Life in the inner city and socio-economic marginalization are not the only factors that create challenges and vulnerabilities for refugee young people. The successful integration of refugee families into Canadian society is essential to the future success and opportunities available to them. However, settlement issues and discrimination are major stress factors, which prevent the successful integration of refugee families (Shakya, Khanlou, & Gonsalves, 2010; Stewart, 2011). The youth of new refugee families who struggle to integrate into Winnipeg are at risk of marginalization and alienation making them more vulnerable to becoming involved with the criminal justice system (Rossiter & Rossiter, 2009). The importance of successful social integration and the positive

influences of families and communities on young refugees is underscored in the practices of the Winnipeg street gang the *Mad Cowz* who have reportedly recruited refugee youth as young as 10 years old into their street gang organization (Macdonald, 2007).

Educational barriers such as English literacy and inappropriate grade placement based on age rather than competency have a significant impact on young people's feelings of depression, isolation and in not fitting in locally (Kanu, 2008; Macdonald, 2007; Neumann, 2011; Shakya et al., 2010). In particular, literacy and language skills are important factors in self-direction, social mobility, and integration, as well as providing opportunities and success in school (Carter et al., 2006). Conversely, those who are unable to develop English and/or French literacy skills are found to be disproportionally represented in the criminal justice system (Rossiter & Rossiter, 2009).

Past experiences of extended periods of disrupted schooling in regions where there is continuous warfare makes it difficult for refugee youth to adapt to and to keep up with expectations of the new host country educational system (Kanu, 2008; Stewart, 2011). These barriers create increased stress among young refugees who are already coping with the stress and trauma of their past experiences as well as the stress of adapting to a new country and culture (Kanu, 2008; Neumann, 2011). Young refugees who are unable to adapt within the educational system may end up dropping out of school. They are thus at a disadvantage for future employment and will most likely face further socioeconomic marginalization (Carter et al., 2006). Experiences of discrimination based on race, religion, language and culture also account for the school drop-out rates of refugee youth as well as their continued feelings of social alienation in society (Kanu, 2008; Neumann, 2011). Further, young refugees who do drop out of

school are at a higher risk of becoming involved with the criminal justice system (Stewart, 2011).

It is these cumulative experiences of discrimination and feelings of isolation that attract refugee youth to criminal activity and antisocial behavior. A former member of the *Mad Cowz* street gang in Winnipeg explained that refugee youth want to be around others who have shared their same experiences, who know what they are going through and to whom they can relate (Macdonald, 2007). He expressed how difficult it is for refugee youth to ±fit inø to their new society (Macdonald, 2007). The former gang member explained that there is nothing for refugee youth to relate to, no place to go, and nothing to do other than get themselves into trouble (Macdonald, 2007).

It is clear that the challenges faced by refugee youth are numerous and complex. As a society, if we fail to address these challenges the likelihood of a successful transition for refugee youth into mainstream society remains minimal. By understanding how various forms of violence and social change may negatively affect people we gain a deeper insight into the challenges faced by refugee youth and the behavior, which may result from these challenges.

1.6 – Framework of the Study

This thesis is divided in five distinct chapters: Introduction, Literature Review, Methodology, Discussion and Results, and a Conclusion.

Chapter one introduces the study by stating its purpose and explaining the study significance. Further, Chapter one also acknowledges the study limitations by highlighting areas not covered in the study. The opening chapter also provides contextual

information with regards to refugees and their various struggles and challenges living in Winnipeg. Finally, Chapter one outlines the study framework.

Chapter two presents a literature review based on relevant concepts and theories from a variety of disciplines including: psychology, education, politics, peace and conflict resolution studies. The literature review is divided into four sections: Youth and Warfare, Experiences with Direct Physical Violence, Structural Violence, and Basic Human Needs.

Chapter three presents the methodology used in conducting this study. Using a qualitative semi-structured interview approach, this study sought to gain a deeper understanding of the experiences and perceptions of young adult male refugees living in Winnipeg with regards to their challenges, coping strategies and influences.

Chapter four presents the first chapter of results and discussion on the research findings. In particular, this chapter focuses on the key themes that emerged throughout the course of the research that my participants identified as factors that contribute to their gang involvement in Winnipeg. The key themes in Chapter four are organized into six categories: Adapting to School, Perceptions of the Neighborhood, Peer Group, Challenges in the Family, Perceptions in the Ethnic Community, and Perceptions of Law Enforcement. Following these categories will be a section on Key Findings

Chapter five presents a second chapter of results and discussion on the research findings. The key themes of chapter five focus on factors relating to young refugeesø positive experiences and point to the respondentsø ideas about how to find a way out of gang life. The key themes in chapter five have been organized into four categories: Influences and Role Models, Turning Points, Coping Strategies, and Addressing the

Issues. As well, a Key Findings category was added which highlights and discusses the key overall findings of the research.

Chapter six is the conclusion which summarizes the thesis. This chapter ties the key findings back to the literature while also exploring identified gaps in the research. Further, chapter six includes recommendations for future research within the context of young refugees and gangs which are beyond the scope of this thesis.

1.7 - Conclusion

This chapter has introduced the study and has provided an outline for the study and shown how this thesis is to proceed. As indicated, the following chapter, chapter two provides the theoretical foundation, which frames the study.

Chapter Two – Literature Review

2.1 - Introduction

A theoretical framework provides structure for the analysis of the violent life experiences of refugee youth and the dramatic social change they have undergone. Further, this framework assists in explaining their perceptions based of their lived experiences. The literature and theory are organized according to four main areas. The first section explores the experiences of youth in war. The second section examines the potential impact of direct physical violence on refugee youth. Third, Johan Galtunges (1969) theory of *structural violence* is explored. This section is followed by a discussion of John Burtonøs (1990) basic human needs theory. This knowledge is synthesized in a discussion about how these concepts inform our understanding of young refugeesø identity and behavior. Our understanding of how exposure to various forms of violence be they physical or structural and one a ability to satisfy basic needs impacts on identity and behavior provides this study with a solid foundation and an informed lens through which to examine how young refugees in Winnipeg respond to these experiences. This approach also provides an understanding as to why young refugees may become involved in gangs, and what motivates them to leave gangs.

2.2 - Youth and Warfare

Many refugee young people have witnessed the massacre of family or community members during the chaos of war (Beah, 2007; Hyder, 2005; Kaplan, 2009; Macdonald, 2007; Wilkinson, 2002). As well, many have been forced to carry out violent acts as

child soldiers (Beah, 2007; Cairns, 1996; Hyder, 2005; Kanu, 2008; Kaplan, 2009; Wessells, 2006) while others are victimized through sexual violence (Kanu, 2008; Kaplan, 2009; Wessells, 2006). Garbarino, Kostelny, and Dubrow (1991a) put it bluntly by stating that õreal war is the end of childhoodö (p. 8)

However, Wessells and Jonah (2006) explain that although youth are often forced into becoming soldiers, there are also cases where youth have voluntarily joined armed struggles. For example, in places like South Africa, Palestine, and Northern Ireland, youth have willingly participated in political violence (Cairns, 1996). As Michael Wessells (2006) explains:

Children who grow up in war zones might not see any positive place for themselves in society; in their situations they are oppressed, have little or no access to education, feel powerless and alienated, and have been denied positive life options. As a result, they may see violence as an acceptable way to replace the existing social order with one offering social justice and positive economic and political opportunities. Finding meaning in the struggle for liberation and social transformation, they may be drawn into armed groups without explicit coercion (p. 3).

Children affected by war may also join armed groups to fulfill a need for family, respect, belonging, and to gain access to valuable resources such as food and shelter, which they may otherwise be denied in ÷civilian lifeø (Wessells, 2006). In fact, joining an armed group may be õthe only means of meeting basic needsö for youth in a war zone (Wessells & Jonah, 2006, p. 33). Further, Wessells (2006) and Wessells and Jonah (2006) explain that some youth join armed struggle out of a sense of responsibility to defend their families and communities. In the violence of war many youth lose one or both of their parents (Wessells, 2006; Wilkinson, 2002). The loss of a family member leads some youth to take up armed struggle in order to avenge the death of that family member (Wessells, 2006).

War as Garbarino et al. (1991a) explain is more than killing and maiming. War deeply affects the social and economic fabric of a society (Garbarino et al., 1991a). The destruction of infrastructure and loss of income for families in a war zone increases poverty levels and prevents youth from accessing adequate food and healthcare (Beah, 2007; Hyder, 2005; Wessells, 2006). Wessells (2006) asserts that war can negatively affect the õphysical, cognitive, emotional, social and spiritualö well-being of youth (p. 23). Hyder (2005) agrees that the violence of war can have a behavioral, spiritual, and psychological impact on youth. Youth are also vulnerable to increased violence in the household as the stresses of living in a warzone manifest themselves through family violence (Wessells, 2006). In spite of these challenges Wessells (2006) argues that youth in war zones have shown resilience where support mechanisms such as õsocial networks, kinship ties, traditions, teachers and religious leaders, elders, and local womenøs associations and youth groupsö (p. 29) were maintained.

In some cases, youth have not only shown resilience in times of war, but they have been active in advocating for peace. For example, Helsing, Kirlic, McMaster, and Sonnenschein (2006) note young Israeli conscientious objectors such as Haggai Matar, who was among four other young men who õspent almost two years under detention and in jail as conscientious objectors, arguing that they could not serve in an occupation armyö (p. 196). The authors state that nonviolent acts of defiance in the name of peace such as conscientious objection to military service by Israeli young people have inspired Palestinian youth to turn away from violence (Helsing et al., 2006).

2.3 - Experiences with Direct Physical Violence

Direct physical violence can be described as õany form of violence that directly impacts on the human bodyö (McCutcheon, 2004, p. 182). Byrne and Senehi (2012) explain that violence is used as both an action to dominate over a perceived competitor, as well as a response to feelings of fear and insecurity. Bombings, killing, rape, and torture are some of the ways direct physical violence is expressed and experienced (Kaplan, 2009).

Continued exposure to direct physical violence increases the likelihood that youth will not only perceive violence as normal, but they will also likely engage in violence and anti-social behavior in the future (Byrne, 1997; Oppenheimer, Bar-Tal, & Raviv, 1999; Senehi & Byrne, 2006). According to Kanu (2008) and Byrne and Senehi (2012) continued exposure to direct physical violence as a normal part of life desensitizes violence for refugee youth. Violence in various forms of the entertainment media can also desensitize youth and socialize them to accept violence as a normal way of life (Grossman, 1996; Miranda & Claes, 2004). Violence, as Carolyn Nordstrom (1997) argues, is not inherent within people or their communities, but rather is õsocially and culturally constructedí ö (p. 7).

Without positive role models teaching youth non-violent methods of problem-solving, many of them view direct physical violence as an appropriate and a normal method of confronting difficulties in their lives (Bemak & Keys, 2000; Byrne & Senehi, 2012; Coles, 1986). Further, Byrne and Senehi (2012) argue that young people who witness direct physical violence at home will have the opportunity to model it resulting in negative behavior. This argument is validated by Garbarino (2001) and Macdonald

(2007) who both draw direct connections between youth violence, and the direct physical violence they have witnessed either in their homes or communities.

In addition, the trauma associated with violent experiences can have a detrimental affect on the cognitive functions of youth (Garbarino, 2001; Wilkinson, 2002). For example, Garbarino (2001) argues that there is õconsiderable empirical evidence for an association between trauma, violence or neglect and school achievement, including intelligence test and performance (p. 8). Coping with trauma can be very challenging for refugee youth as they experience stress in integrating into and adapting to their new society (Rossiter & Rossiter, 2009; Senehi & Byrne, 2006). Rossiter and Rossiter (2009) also argue that youth who have been exposed to violence and trauma prior to migration are vulnerable and at greater risk of becoming involved in violent criminal activity. As well, many youth may use direct physical violence as a coping mechanism when faced with stressors in their everyday lives (Rossiter & Rossiter, 2009; Sersli, Salazar, & Lozano, 2010). Instances of direct physical violence carried out by refugee youth highlight the need to fully understand their behavior which at times may not be acceptable in public settings such as the school (Rossiter & Rossiter, 2009; Sersli et al., 2010).

2.4 - Structural Violence

Many refugee young people and their families have experienced various forms of institutional exclusion, racism, or ethnic prejudice in their countries of origin (Cairns, 1996; Stewart, 2011; Suarez-Orozco & Suarez-Orozco, 2001; Wolff, 2006). Sadly, refugee young people and their families continue to experience exclusion here in

Winnipeg with regards to their access to suitable education, opportunities for higher education, adequate housing and employment (Carter, 2009a, 2009b; Carter et al., 2006; Kanu, 2008, 2009; Stewart, 2011). By understanding the role and impact of structural violence we are provided with a more comprehensive understanding of the experiences refugee young people have with different forms of hidden violence, social change and exclusion.

Unlike direct physical violence where one person or group causes direct harm, or pain to another person or group, structural violence is indirect in that there is no one person causing physical pain or hurt upon another (Galtung, 1969). Rather, structural violence is embedded within the very structure of society institutions and is identified as unequal power relations, unequal opportunities, and unequal distribution of, and access to resources (Galtung, 1969; Byrne & Senehi, 2012). Galtung (1969) stresses the importance of recognizing the stratification of society class structure to fully understand structural violence. Moreover, Byrne and Senehi (2012) argue that in order to better understand the causes of violence the structural conditions of society must also be understood. They assert that violence is often the result of social divisions based on race, power, ethnicity, class, culture, gender, sexual orientation, religion, as well as other identities and contrasting ideologies.

According to Galtung (1969) actors within the system interact in a hierarchical order so that the higher one is in the system, the more central that personos position of power is within that system. Those who hold a lower position within the system are deprived of realizing their potential (Galtung, 1969). For Galtung (1969) oviolence is present when human beings are being influenced so that their actual somatic and mental

realizations are below their potential realizationsö (p. 168). In this sense, violence is õthe cause of the difference between the potential and the actual, between what could have been and what isö (p. 168). For example, the denial of starving people access to food when in fact there is enough food to feed them (Galtung, 1969).

Arendt (1970) argues that rage and violence will arise if people's expectations that oppressive conditions could change in fact never happens. ÕDifficult life conditions make people feel threatened as they are unable to care for themselves or their family (Staub, 1989; Byrne & Senehi, 2012). These threats as Staub (1989) asserts, can be both physical as in the protection of one's own life, and psychological as in the protection of one's values, self-esteem, and identity. However, õdifficult life conditions may not in and of themselves lead to violence (Staub, 1989). Staub (1989) argues that õhostility is especially likely to arise if people regard their suffering as unjust, as they often do, and especially if some others are not similarly affected (p. 16).

2.5 - Basic Human Needs

Closely associated with the concept of structural violence is John Burtonøs that basic human needs deprivation constitutes violence. Basic human needs include physical needs such as: food, clothing and shelter, and also include self-esteem, love, belonging, opportunity, security, employment, and freedom from fear, violence and marginalization (Byrne & Senehi, 2012; Jeong, 2000; Rossiter & Rossiter, 2009). These needs are universal and cannot be negotiated (Burton, 1993; Byrne & Senehi, 2012; Jeong, 2000; Roy, 1990). Basic human needs can be accessed and satisfied through human interaction and participation in community life (Jeong, 2000).

Unequal opportunities and a corresponding inferior social status can prevent people from satisfying their basic needs (Jeong, 2000). The literature cites several significant challenges refugee youth face in realizing their basic needs: family instability, inferior socioeconomic status, poor physical and mental health, lack of language proficiency, inadequate social skills, lack of social networks, inadequate employment opportunities, racism, and neighborhood settlement that are all barriers to satisfying their basic needs (Rossiter & Rossiter, 2009; Sersli et al., 2010). Further, Neufeld-Redekop (2002) argues that the satisfaction of needs, which seek to fulfill omeaning, connectedness, action, security, and recognition have been shown to be important in developing identity for the human Selfo (p. 164). When these needs are satisfied a sense of well-being is achieved that is built upon oself-recognizance, self-respect, self-esteem, self-confidence, and self-actualizationo (Neufeld-Redekop, 2002, p. 164-165).

Failure to satisfy basic needs may result in õanger, sadness, depression, fear, and shameö (Neufeld-Redekop, 2002, p. 165). As well, the failure to fulfill young peopleøs basic needs may breed resentment and violence as they unsuccessfully struggle against the structure to satisfy those needs (Burton, 1993; Byrne & Senehi, 2012; Galtung, 1969; Gilligan, 1996; Jeong, 2000). Thus, hopelessness, lack of opportunity and feelings of social injustice make gang involvement an attractive option for disadvantaged youth (McCluskey & Torrance, 2003; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wortley & Tanner, 2007). Refugee youth may join gangs for social and psychological benefits as gang membership may meet unfulfilled needs such as: economic benefits, respect, dignity, identity, belonging, opportunity and security (Byrne & Senehi, 2012; Sersli et al., 2010; Wortley & Tanner, 2007).

Basic human needs theory rejects the notion that violence is embedded within the very nature of human beings (Jeong, 2000). Gilligan (1996) also rejects the notion of an inherent evilness within human nature. Rather, Gilligan (1996) argues that violence is a public health issue whereby the causes of the illness (in this case violence) are examined in order to cure the illness and prevent its spread. Through a similar lens Prothrow-Stith and Spivak (2004) explain youth violence as an õepidemicö in which violence is a result of ocomplex problemso that require ocomplex solutionso. Further, Garbarino (2001), Vorrasi and Garbarino (2000), and Howard and Jenson (1999) also argue that the advent of youth violence is not due to a singular cause or challenge, but rather is the result of the accumulation of multiple risk factors. Where multiple risk factors exist, Garbarino (2001) states that the prevalence of youth violence õi tends to be overwhelmingö (p. 173). Prothrow-Stith and Spivak (2004) argue that it is critical for society to pay attention to these risk factors in order to reduce youth violence. Acknowledging the ocomplex problemso and ocomplex solutionso notion put forth by Prothrow-Stith and Spivak (2004) Byrne and Senehi (2012) state that any youth violence and youth gang prevention/intervention approach omust engage at all levels to transform the underlying nature of youth conflict over timeö (p. 62).

An examination of the literature reveals the potential for challenges and risk factors for refugee youth in four key areas:

The school (Bemak & Keys, 2000; Byrne, 1997; Byrne & Senehi, 2012; Davis, 1999; Howard & Jenson, 1999; Kanu, 2008, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wilkinson, 2002; Wortley & Tanner, 2007)

- The family (Bemak & Keys, 2000; Byrne & Senehi, 2012; Byrne, McLeod, & Polkinghorn, 2004; Davis, 1999; Howard & Jenson, 1999; Kanu, 2008, 2009; Kaplan, 2009; Molix & Bettencourt, 2010; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wilkinson, 2002)
- The community (Bemak & Keys, 2000; Byrne & Senehi, 2012; Byrne et al., 2004; Chekki, 2006; Davis, 1999; Howard & Jenson, 1999; Kanu, 2008; Molix & Bettencourt, 2010; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wortley & Tanner, 2007); and
- The peer group (Byrne & Senehi, 2012; Davis, 1999; Howard & Jenson, 1999;
 Kanu, 2008, 2009; Molix & Bettencourt, 2010; Neumann, 2011; Rossiter &
 Rossiter, 2009; Sersli et al., 2010; Wortley & Tanner, 2007)

These social institutions õí shape young adults behavior as they acquire the knowledge and skills to participate effectively in the communityö (Byrne, McLeod and Polkinghorn, 2004, p. 15).

2.5.1 The School

Education is essential for refugee young people to have a smooth transition into a new society and so that they can access opportunities (Rossiter & Rossiter, 2009; Wilkinson, 2002). Refugee young people whose schooling was interrupted, and who lack the necessary language proficiency are at a disadvantage for completing high school, and also for their employability after graduation (Kanu, 2009; Rossiter & Rossiter, 2009; Wilkinson, 2002). In their study of disadvantaged Aboriginal youth within the school system in Winnipeg McCluskey and Torrance (2003) found that many youth simply find

the extra challenge of school too much to cope with and end up dropping out. Similarly, Rossiter and Rossiter (2009) discovered that 46-74 percent of immigrant and refugee youth whose native language is not English fail to complete high school.

Unfortunately, schools and teachers often lack the experience and understanding, as well as the necessary resources to meet the unique needs and challenges of refugee young people (Kanu, 2008; Rossiter & Rossiter, 2009; Sersli et al., 2010). Without the appropriate training for teachers and the much needed resources, the behavior of refugee young people is often misunderstood, and as a result they may be labeled as #rouble makers@ This label and the stigma that comes with it may push young people into association with other similarly labeled youth effectively increasing their risk for gang involvement or other criminal behavior (Byrne & Senehi, 2012; Kanu, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wortley & Tanner, 2007).

Schools have the ability to play a significant role in the socialization of young people through which their understandings and perceptions of the world are informed and shaped constructively to empower them (Byrne, 1997; Senehi & Byrne, 2006; Sersli et al., 2010). Further, of the way that young people, therefore, understand their interpersonal and intergroup relations within their sociocultural milieu impacts their perceptions of conflict, violence, and peace, and conflict management stylesö (Polkinghorn & Byrne, 2001, 25). Jay MacLeod (2009) argues that schools need to rethink their role in the social reproduction of the class system through the way they value the cultural capital of students, which tends to favor the oupper classesö. As MacLeod (2009) states, of the assumption of some mainstream sociologists that the problem must lie with the contestants, rather than with the judge, is simply unfoundedo (p. 101). In other words,

MacLeod is suggesting that schools need to reexamine the way they educate to better suit the needs of the students, rather than forcing students to fit into a particular banking system of educating.

In order to reduce the risk factor, refugee young people need to be engaged and empowered at school through sports, tutoring, culturally appropriate teaching and mentoring, while also learning problem-solving and anger management skills (Byrne & Senehi, 2012; Byrne et al., 2004; Kanu, 2009; McCluskey & Torrance, 2003; Sersli et al., 2010). The appropriate educational inputs to engage with refugee young people will provide them a greater sense of self-esteem and a more positive outlook on education making them more likely to be successful in various avenues of life (Rossiter & Rossiter, 2009; Sersli et al., 2010).

2.5.2 – *The Family*

Many refugee families settling in a new country continue to face difficult life circumstances, which may present challenges and substantial factors for refugee youth. As well, the parents of refugee youth themselves are often dealing with trauma and struggles such as finding suitable employment to support the family (Rossiter & Rossiter, 2009; Sersli et al., 2010; Shakya et al., 2010). In many cases refugee parents may have to work more than one low-income job as they often do not have adequate education, language proficiency, or professional skills that would empower them to obtain a career (Rossiter & Rossiter, 2009; Sersli et al., 2010). The need to work multiple jobs especially in single parent households pulls parents away from the family and leaves young people to supervise themselves putting them at risk of becoming involved in

crime, gangs and violence (Byrne & Senehi, 2012; Davis, 1999; Kanu, 2008, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Stewart, 2011; Wortley & Tanner, 2007).

The struggle to meet the basic needs of the family may require refugee youth to also seek employment, which could negatively affect their performance and attendance at school (Kanu, 2008, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Stewart, 2011). The risk of failure becomes greater when parental attitudes towards education are negative or indifferent (Byrne & Senehi, 2012; Rossiter & Rossiter, 2009; Sersli et al., 2010). The drive to make money may attract some youth to criminal activities where money may be more lucrative and faster to earn (Rossiter & Rossiter, 2009; Stewart, 2011).

Other family risk factors include the adjustment gap that develops between refugee youth and their parents. Children learn English with more ease than their parents, which prevents parents from assisting their children with after school work (Kanu, 2009; Rossiter & Rossiter, 2009; Wilkinson, 2002). As well, strain is created between parents and children as parents tend to hold onto traditional views and values while their children are more exposed to the influences and values of their peers and the society around them (Chekki, 2006; Kanu, 2008, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wilkinson, 2002).

Sustained support mechanisms such as skills training and support groups for refugee parents and families throughout the settlement process are essential for diminishing the risk for youth involvement in criminal activity (Rossiter & Rossiter, 2009; Sersli et al., 2010; Wortley & Tanner, 2007). The importance of family support programs is highlighted by Sersli et al. (2010), and Wortley and Tanner (2007) who

indicate that it usually takes a few years of struggling before refugee youth succumb to gang involvement. Byrne, McLeod and Polkinghorn (2004), and Molix and Bettencourt (2010) assert that when families are able to offer the necessary support to their children, it is more likely their children will show resilience when faced with violence and other challenges.

2.5.3 – The Community

Community environment both in terms of geography and culture plays a significant role in the level of risk associated with refugee youth. Due to the socioeconomic challenges that many refugee families face they often reside in low-income and high crime neighborhoods where they lack the resources of other wealthier neighborhoods (Rossiter & Rossiter, 2009; Sersli et al., 2010). Studies conducted in the United States found that the majority of youth who commit violent crimes also live in low income inner-city neighborhoods (Vorrasi & Garbarino, 2000). Living in these crime ridden and impoverished neighborhoods puts refugee youth at risk of gang involvement, as well as exposing them to violence (Byrne & Senehi, 2012; Sersli et al., 2010; Wortley & Tanner, 2007). Thus, there is a correlation between the neighborhoods where ethnic minority youth live and the level of gang activity occurring in these areas (Neumann, 2011).

Community organizations need effective programming that promotes training, education and social capital for refugee youth and families (Chekki, 2006; Sersli et al., 2010). As well, culturally appropriate trauma reduction programs and resources for

refugees dealing with trauma need to be promoted and be accessible as many refugee families report not receiving any trauma related treatment (Kanu, 2008).

The existence of cultural and faith-based institutions can also provide a stabilizing factor for refugee youth (Chekki, 2006; Sersli et al., 2010). For example, Rossiter and Rossiter (2009) found that those refugee youth who had a strong sense of cultural identity were at less risk of becoming involved in illegal activities and showed a greater interest in school. Further, they were also found to have superior decision-making skills. Chekki (2006) also asserts that cultural and ethnic organizations can help meet material, and spiritual needs as well as satisfy the need for belonging. Further, communities lacking these types of organizations may lack the leaders and positive role models necessary to steer youth in the right direction (Rossiter & Rossiter, 2009). Without positive adult role models to assist refugee youth in feeling secure, some youth may turn to gangs to gain a sense of safety (Rossiter & Rossiter, 2009). Byrne and Senehi (2012) also stress the importance of community leaders such as elders who can offer positive values and ideas while nurturing young people in the community.

2.5.4 – The Peer Group

Refugee youth who struggle with feelings of isolation, low self-esteem, discrimination, exclusion and loneliness often are attracted to other disenfranchised youth and become vulnerable to negative influences resulting in antisocial behavior (Kanu, 2008, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Wortley & Tanner, 2007). Byrne, McLeod and Polkinghorn (2004) state that ofthe individualos desire to belong to a particular social group is usually motivated by a desire to enhance self-esteemö (p. 16).

Rossiter and Rossiter (2009) also argue that refugee youth face pressure to fit-in and gain acceptance from others. When the schools, family, and community fail to meet the needs of refugee youth, local gangs are ready to fill the gap (Sersli et al., 2010). For example, Sersli et al. (2010) point out that youth who join gangs feel a sense of empowerment, security, belonging, and are able to access economic opportunities that they are unable to receive through other institutions. To counter the attraction of gang involvement, Sersli et al. (2010) argue that support is needed for recreational programs that get refugee youth involved in their communities and provide them with positive sports mentors.

2.5.5 - Identity/Behavior

Refugee youth who experience life in a new country are faced with an identity crisis as they struggle with fitting into a new culture and a new value system (Rossiter & Rossiter, 2009; Sersli et al., 2010). The resulting oculture shocko leaves refugee youth straddling two worlds and not fully belonging to either as they walk in between (Rossiter & Rossiter, 2009). Fisher (1990) stresses the importance of identity when he states that of identity is a fundamental need that influences a great deal of social interaction at the group, intergroup and international levels.ö (p. 95).

Research shows that minority groups are often devalued and experience stigmatization and discrimination within mainstream society (Molix & Bettencourt, 2010). Feelings of isolation and exclusion feed into negative stereotypes held by both minority and dominant groups (McCluskey & Torrance, 2003). McCluskey and Torrance (2003) argue that as minority groups search for their identity they often take on stereotypes, which may lead them to express themselves through antisocial behavior such

as participating in illegal activities. Suarez-Orozco and Suarez-Orozco (2001) refer to this behavior as *Negative Social Mirroring* whereby identity and behavior is shaped by how others perceive us and how they project that image back on to us as we then take on that identity. The authors state that as frustration grows within these youth they can begin to react with anti-social behavior. õFor these youth, the response is ÷you think Iøm bad? Let me show you how bad I can beøö (Suarez-Orozco & Suarez-Orozco, 2001, p. 100). Similarly, when young people live in the inner city they may feel the need to reflect a õtough guyö image in order to gain respect from their peers and others in the community (Byrne & Senehi, 2012; Howard & Jenson, 1999).

Young peopless perceptions and experiences of exclusion, discrimination, and oppression shape their identity and are passed along through storytelling (Jeong, 2000; Senehi & Byrne, 2006; Sersli et al., 2010). The Rwandan Genocide is a well known example of how young peopless identity was shaped through storytelling. In the lead up to the genocide poets, singers and school teachers told stories harkening back to historical conflicts between the Hutu and Tutsi people (Batiskunda, 2009). The storytelling of past wrong doings was so successful Batiskunda (2009) notes that õmost of the killers during the genocide were people who were not even born before the 1959 Hutu Revolutionö (p. 149). He added that many of the killers had never personally experienced õanything negative from Tutsis but absorbed the hatred ideology as gospelö (p. 149). As identity becomes entrenched, younger generations take on their group identity and behavior as their own (Byrne & Senehi, 2012; Senehi & Byrne, 2006). For example, Macdonald (2007) found that some refugee children in Winnipegs inner city mimic the behavior of gang members from their community in their play. As children learn this type of

negative behavior, the importance of positive behavioral role models from family, schools, peers and community becomes important to prevent their transition into violence (Byrne & Senehi, 2012).

2.6 – Conclusion

As evidenced by the literature refugee young people face a multitude of risks and challenges, which could negatively affect their behavior and lead them down the path to anti-social behavior and criminal gang activity. However, where positive support mechanisms exist refugee young people have a greater chance of showing resiliency when faced with these challenges. Discovering the challenges and support mechanisms that exist in Winnipeg for refugee young people is important for service providers and young refugees to better understand how these challenges inform their identity and behavior and why they would potentially choose to join a gang or leave a gang once they are members. In order to acquire this information and knowledge my research methodology is outlined in Chapter three.

Chapter Three – Methodology

3.1 - Introduction

Young people's experience as refugees living in a new milieu is a challenge and informs their perceptions, attitudes, and behavior. This study uses a qualitative research method to explore young refugees perceptions, challenges and experiences as new-comers living in Winnipeg. In particular, a semi-structured interview approach was used when interviewing the study participants.

3.2 – Research Location

As noted by Carter et al., (2006) the majority of refugees who settle in Winnipeg, settle in the inner city. Not surprisingly then, this study was conducted in Winnipegs inner city Central Park downtown neighborhood where all the young men participating in this study settled upon arriving in Winnipeg.

Home to people representing over fifty countries (Carter, 2009a), the inner city of Winnipeg is a culturally and ethnically diverse area of the city. The Central Park neighborhood in particular is one of the most ethnically diverse neighborhoods in Canada (Kirbyson, 2012). Appropriately, many immigrant and refugee service organizations such as Immigrant Centre, Manitoba Interfaith Immigration Council, Immigrant Refugee Community Organization of Manitoba, and Newcomer Employment Education Development Services are all located within or near the Central Park neighborhood. As well, many of the small grocery stores, clothing stores and restaurants in the area reflect the ethnic diversity of the neighborhood. Moreover, Central Park itself becomes

transformed into a global market on summer weekends where customers can purchase everything from handicrafts to food that reflects the diversity of the neighborhood.

Although Winnipegs inner city is the most affordable area of the city to live in (Carter et al., 2006), it also has had its share of challenges. Characterized by high levels of poverty (Carter, 2009a), and õunsuitable, overcrowded and unsafe housingö (Carter et al., 2006, p. 53) Winnipegs inner city is marginalized. Not only do the residents of the inner city face high levels of poverty along with inadequate housing, as Fitzgerald, Wisener and Savoie (2004) note, the inner city, and particularly the downtown area of Winnipeg where the Central Park neighborhood is located, has the highest rates of violent crime and property crime in the city of Winnipeg.

Recently the Central Park neighborhood has undergone some positive changes. Thanks to a 5.5 million dollar development initiative, Central Park is transforming from an area once infamous for its crime and drug use to a family friendly area hosting a toboggan slide, soccer pitch, and the cityøs largest spray park (Kirbyson, 2012). This influx in community development has contributed to a fifty percent decline in crime for the area over the past few years (Kirbyson, 2012). However, outspoken *Winnipeg Sun* columnist Tom Brodbeck (2013) questions these statistics claiming that city officials are redrawing boundaries and cherry picking their statistics. Instead Brodbeck claims that crime has actually gone up in the downtown area by 13.5 percent (Brodbeck, 2013). Regardless of these statistics it remains clear that although the Central Park neighborhood has experienced some positive developments recently, the residents of this neighborhood still face multiple challenges.

3.3 Qualitative Research Strategy

The use of qualitative interviews can greatly expand the depth of information collected in the field (DiCicco-Bloom & Crabtree, 2006). Qualitative interviews are different from the very rigid and structured survey type research and tend to be less structured, more flexible, and more focused on the interviewee's experiences and points of view than the very rigid and more structured survey type research (Bryman & Bell, 2007). For example, Druckman (2005) notes that one of the advantages of a qualitative research method over a quantitative research method is that it can oprovide a deeper understanding of a phenomenonö (p. 8). As well, Druckman (2005) points out that a qualitative methodology empowers the researcher to identify nuances which are missed in quantitative methodology. Bryman and Bell (2007) also support this point explaining that by providing the opportunity for the interviewee to speak freely deeper insight is given to what he or she feels is relevant and meaningful for a particular topic. Further, Bogdan and Knopp Biklen (2003) note that the open-ended responses allowed for in qualitative research provide the researcher with an opportunity to collect õunexpected dimensions of the topicö from the participant (71). However, Johnston (2005) warns that othere needs to be balance between what the researcher wants to ask about and what the interviewee wants to talk about; the balance will dictate what is important in the interviewö (p. 282).

Smith (1999) asserts that by providing space for people to tell their stories many individual truths can be revealed. This constructivist method seeks to uncover the multiple perceptions and experiences of the study participants (Druckman, 2005). Consequently, Johnston (2005) explains that in narrative analysis it is not the researcher¢s

goal to õvalidate the accurateness of the personøs story, but rather to discover the meaning of itö (p. 282). Lederach (2005) stresses that when we listen to peopleøs stories we must look for õdeeper social story and meaningö (p. 147). When we listen to peopleøs stories and gain a deeper understanding of their perceptions and experiences we can also begin to understand how their behavior has become influenced by these perceptions and experiences (Johnston, 2005).

Johnston (2005) states when people tell their stories, they are the experts. This process of expression allows marginalized people in particular to dictate how they will be represented in society, thus providing a counter-image to the dominant discourse (Smith, 1999). In addition, Senehi (2009) explains that õwhen only those in power have access to producing knowledge, authoritative-discourses may serve the interests of power rather than truthö (p. 203-204).

If we truly want to address the violence and marginalization the young people in Winnipegos inner city are experiencing, it is imperative that concerned persons seek out and listen to their stories because they live through these experiences on a daily basis. During the research process I sought to follow this approach.

3.4 - Semi-Structured Interviewing

Unlike structured interviews in which õinterviewers must ask questions in the exact order and with the exact wording in which they are written in advanceö (Rubin & Babbie, 2011, p. 233), in semi-structured interviewing the researcher generally has an interview guide outlining the topics and issues, which should be covered during the course of the interview. This less regimented approach allows the interviewer to be

õflexible, informal, conversational and to adapt the style of the interview and the sequencing and wording of questions to each particular intervieweeö (Rubin & Babbie, 2011, p. 233). This flexibility within the interview process allows the interviewer to ask new questions as deeper insights are provided during the course of the intervieweeøs sharing of their experiences and stories that capture their perceptions of events (Bryman & Bell, 2007; DiCicco-Bloom & Crabtree, 2006). DiCicco-Bloom & Crabtree (2006) state that the goal of the interviewer õis to encourage the interviewee to share as much information as possible, unselfconsciously and in his or her own wordsö (p. 316-317). Further, Barriball and While (1994) point out semi-structured interviewing allows for the exploration of the õperceptions and opinions of respondentsö (p. 330). As interviewees share their experiences and perceptions with the interviewer, probing questions can elicit valuable information and make for more complete and informed research (Barriball & While, 1994).

3.5 - Role of the Researcher

The purpose of using a semi-structured interview technique is to allow participants to freely share their own perceptions, experiences, and stories with the researcher. Consequently, I took on the role of independent interviewer.

Within the flexibility of the semi-structured interview process it was my intent to elicit stories from my study participants. This is not to say that semi-structured interviews and storytelling are one and the same, rather stories can unfold within semi-structured interviews as the participant shares his or her life experiences with the researcher. This participatory process humanized the research õsubjectö as an active

participant in the research process by providing a space for my participants to share their experiences and stories with me. Storytelling, as Senehi and Byrne (2006) point out, is a way to empower youth as they can express their own experiences and perceptions while also having the power to choose what they share, and how much they share. It is the sharing of these stories that creates a better understanding of young peopless perceptions and circumstances. For example, Senehi (2009) states that õstorytellers often invoke the past to comment on problems and needs in the present in order to affect the futureö (p. 203). Smith (1999) points out that storytelling is often a way for indigenous and other marginalized groups to voice their struggles and challenges, yet it is also an opportunity to determine their own solutions without the ipaternalismo of government or nongovernmental organizations.

Lederach (2005) asserts that when people are prevented from telling their stories narrative becomes broken. Moreover, he states that when the narrative is broken õmeaning, identity, and a place in history are lostö (p. 146). These insights help the listener understand the importance, meanings and perceptions attached to certain events or circumstances. In particular, the storytelling process helped me gain a richer and deeper understanding of the participant and how he perceived his life experiences.

3.6 - Data Gathering Techniques

I used face-to-face semi-structured interviews to collect the data for this research study. In my research questions I kept the questions broad so to allow the interviewee to really express his or her own thoughts and ideas. Probing questions were used as a guide to encourage the interviewees to share more deeply about their experiences, perceptions,

and challenges (see Appendix A). I purposely avoided using direct questioning regarding violent events, which may have been experienced by some of the study participants. Bogdan and Knopp Biklen (2003) assert that in qualitative research, a good interview õtakes on a shape of its ownö (p. 95). Therefore, it was my hope that as each participant felt comfortable in the interview process, his or her personal experiences and challenges would emerge in the stories that he or she shared.

All of the interviews except for one were recorded on a digital audio recorder and then transferred to a computer hard-drive from where they were transcribed verbatim and analyzed by me. The importance of audio recording during the interview process is stressed by Bryman and Bell (2007), and Barriball and While (1994) who also note that audio recording ensures that the interviewer captures the exact words of the interviewee. Furthermore, Barriball and While (1994) state that audio recording reduces the potential for error in transcribing interviews. The sole interview that I did not audio record at the participant was instead recorded using hand written notes. For each participant the interview process took approximately 60 to 80 minutes depending on the depth to which each participant shared his or her stories with me.

3.7 - Data Analysis

Using a Grounded Theory approach, I used inductive analysis as themes emerged from the data. Grounded Theory refers to the õdiscovery of theory from dataö (p. 1). Using this approach means that õmost hypotheses and concepts not only come from the data, but are systematically worked out in relation to the data during the course of the researchö (p. 6). Data analysis refers to the õprocess of systematically searching and

arranging the interview transcripts, fieldnotes, and other materials that you accumulate to enable you to come up with findingsö (Bogdan & Knopp Biklen, 2003, p. 147). As various themes and patterns emerged throughout the data analysis, a coding system using colored pencils was used to identify each theme. Once the coding of the data was completed, significant thoughts, ideas and quotations were identified to be used in the two discussion and results chapters of this thesis. Further, within the two results and discussion chapters of this thesis, the literature was referenced in order to give a deeper understanding and analysis of the various findings within the data.

3.8 - The Study Participants

This study has a total of eleven participants. Four of the participants are young males aged 19-27 who settled in Winnipeg as refugees. In particular, this study sought out young male refugees who had previously been members of gangs, and who have since left the gangs and have made positive changes in their lives. These young men were selected in order to gain a better understanding of young male refugeesøchallenges, experiences, coping strategies and turning points as new-comers.

Three out of the four young men are former members of a Winnipeg street gang and have all spent time in prison. Although none of the men are currently behind bars, two of them have deportation orders filed against them and if they fail to meet their court ordered probationary terms, they may be deported. The fourth young man was never an official gang member, yet he nonetheless associated with gang members and participated in anti-social behavior.

Three of the four men are from countries in East Africa and one of the men is from a country in West Africa. The refugee experience of all four of these young men varied: only one participant spent time living in refugee camps, another participant fled with his family when he was still a child and settled in a city of a neighboring country, and the other two participants were both born in neighboring countries after their parents fled their country of origin and thus they have never visited their country of origin. Three of the four men came to Canada when they were thirteen years old and one of the four men came to Canada when he was seventeen. The length of time each participant has resided in Canada ranges from 4 years to 15 years. These four young men were given the pseudonyms Bashir, Ermias, Okot, and Emmanuel to protect their identity.

In order to provide a more informed and in-depth analysis of the various challenges and experiences encountered by young refugees, this study also includes five community worker participants, three of whom are male, and two are female. Although the focus of this thesis is on the perspectives and experiences of young male refugees living in Winnipeg, the community worker participants also added valuable data as they shared their own personal experiences and stories about refugee young people in Winnipeg. All three men had themselves originally come to Canada as refugees and were from East African backgrounds whereas both females were both born and raised in Canada and come from European backgrounds. All five community workers work for organizations, which provide programming and services to refugee young people and their families. The three male community workers were given the pseudonyms Hassan, Sadiiq, and Saalim; the two female community workers were given the pseudonyms of Jennifer and Heather.

As the interviews unfolded and participants shared with me some of their challenges, two particular themes emerged from the interviews which spoke to the perspectives participants held of the local police force as well as leaders in their respective ethnic communities and the refugee community in general. As a result of these emerging themes, an officer from the Winnipeg Police Services who has experience working with gang-involved refugee youth was added to the study as a participant. Further, one community elder representing one of the East African communities was also selected to participate in this study. The community workers, the police officer and the elder added valuable data to the study because they shared their own experiences about working with refugee young people in Winnipeg. The community elder came to Winnipeg as a refugee and is from the same ethnic community as one of the young male participants in this study. The community elder is an active leader in his ethnic community and has many years of experience working with his community so youth and their parents.

The Winnipeg Police Services officer was simply given the pseudonym Officer Smith, and the community elder was given the pseudonym Abraham. I thought it was important to add these two additional participants as the young men interviewed, as well as some of the community workers in this study had strong feelings in these two areas. Therefore, it was necessary to include a voice from the police as well as from an elder in order to gain further insight about how they perceive the challenges and issues regarding refugee young people within the city of Winnipeg.

In the original proposal I stated that a mix of male and female young adult refugee participants were going to be selected for this study. However, as the research

process developed it became evident that finding suitable female participants was going to be a real challenge. Several attempts were made at meeting with two different potential female participants. However, after they both failed to attend several scheduled, and then rescheduled meetings, it was decided to refocus on male participants as I had access to that group.

The study participants had sufficient understanding of the English language so I did not need interpreters. All of the study participants were selected based on recommendations from three key informants. Two of these key informants were community workers who participated in the study as such. Having either worked with or come from the same ethnic community as my refugee young adult participants, my key informants had a preexisting relationship with my participants, and therefore were aware of their status as ex-gang members or former gang associates. Further, my key informants were able to direct me to the community worker participants as well as the community elder participant, and the Winnipeg Police Services officer participant. The key informants were identified through preexisting relationships I have with members of the refugee community, as well as a preexisting relationship I have with staff members of the Immigrant Refugee Community of Manitoba (IRCOM). IRCOM is a community based refugee serving organization located in Winnipegos inner city where I have spent considerable time as a volunteer. Having an established relationship with the key informants allowed me to rapidly build relationships with the participants. Further, seeing as the participants also had trusting relationships with the key informants, this provided me with the credibility and trust needed to engage with these young people. As Wilson (2008) notes, $\tilde{o}(s)$ hared relationships allow for a strengthening of the new

relationship. This allows you to become familiar or comfortable with the personö (p. 84). I used the snowballing sampling technique with my young male participants whereby they would use their social networks to refer to me other potential and suitable participants for the study (Luborsky & Rubinstein, 1995). However, this was unsuccessful due to the suspicion and caution the friends of my young male participants held toward me.

I applied for and received human ethics approval from the University of Manitoba for the project prior to contacting potential study participants. All of the participants were provided with a consent form prior to the interview informing them of the purpose of the study and the topics to be discussed. All participants in this study are at least 18 years of age and all of the participants have lived in Winnipeg for a minimum of four years. The actual names of the study participants and their countries of origin will remain confidential and do not appear in the transcripts and will not appear on any reports based on the research. As noted earlier, in place of their actual names, pseudonyms are used to identify the participants within the research. The use of pseudonyms effectively addressed any concerns over identifying and exposing a participant in cases where they shared information with me that may be sensitive.

Throughout the duration of the research all of the transcripts were kept in a locked room in my apartment that was only accessible to the researcher. Digital voice recordings of the interviews were stored on a password protected computer hard-drive only accessible to the researcher. All transcripts were destroyed after the completion of the study.

Following the transcription of the interviews, each participant had the opportunity to receive a transcript of her or his interview. This process provided the study participants with an opportunity to clarify any remarks as well as request that I delete any remarks if they did not feel comfortable keeping them in the transcripts. The importance of accurately representing peopless knowledge the way they it want to be represented is underscored by Smith (1999) when she discusses the misrepresentation of the Maori people within western based research. Smith (1999) states that õ[t]his in turn has entrapped Maori people within a cultural definition which does not connect with either our oral traditions or our lived realityö (p. 170). During the course of the interview process I have not deliberately or knowingly deceived any of the participants.

3.9 – Conclusion

The use of semi-structured interviews and the storytelling which can unfold within them fits well with my own worldview and experience. Having spent several years working with community-based organizations in Uganda and in Winnipeg, I have worked with marginalized members of society including refugees. During this time I have come to understand the importance of listening to the voices of those usually silenced by the dominant society. We can begin to build peace, and we can begin to undertake appropriate and sustainable models of development by listening to these voices. When we have power :withø and not :overø, when we :do withø and not :do toø we can begin to work together to mend the fractures which cripple our societies. This qualitative research study is an attempt to better understand the perceptions, experiences, and challenges of refugee young people living in Winnipeg. It is my hope that through

the stories shared with me by my participants that the information and knowledge will inform future programs and policies aimed at these young people.

Chapter Four – Down the Path to Gang Life

4.1 - Introduction

During the interview process the participants had the opportunity to share their experiences and challenges as well as offer insight and perspectives with regards to the state of young refugees living in Winnipeg. During this process several interesting themes emerged which highlighted the negative factors that contributed to the gang involvement of my participants.

Prior to coming to Canada many refugees tend to believe that their new country will be somewhat of a promised land, where the struggles and challenges of their current situation will be replaced with opportunity and hope for a brighter future. Suarez-Orozco and Suarez Orozco (2001) note for example, that õwhile anticipating migration and during the first weeks in the new country, many experience a sense of euphoria. Expectations are often high, for many the anticipated possibilities seem boundlessö (p. 72). Both Ermias and Emmanuel stated that they thought moving to Canada would provide them with a better education and a better life. Bashir also expressed his expectations upon coming to Canada as follows:

When you see it in the T.V. it different from here. I thought I would live in a house, not an apartment.

As well, Ermias expressed his high expectations prior to arriving in Canada:

I thought Canada was gonna be the best life ya know? It was gonna be like everything I ever wanted, you know what I mean? I thought it was gonna be like heaven!

Sadly however, the opportunities and the promises of a better life all too often remain out of reach for many refugee young people and their families as they now face a new set of challenges such as adapting to a new life, a new culture, and a new society that is quite often very different from their own. When writing about the stresses of immigration Suarez-Orozco and Suarez-Orozco (2001) noted:

Indeed, by any measure, immigration is one of the most stressful events a family can undergo. It removes family members from many of their relationships and a predictable context: community ties, jobs, customs, and (often) language. Immigrants are stripped of many of their significant relationships 6 extended family members, best friends, and neighbors. They also lose the social roles that provide them with culturally scripted notions of how they fit into the world. Initially, without a sense of competence, control, and belonging, many immigrants will feel marginalized. These changes in relationships, contexts, and roles are highly disorienting and nearly inevitably lead to a keen sense of loss (p. 70).

The experiences and challenges shared by the participants in this study reflect the feelings of isolation, marginalization and loss noted by Suarez-Orozco and Suarez-Orozco (2001).

This chapter discusses the results of each of the six themes that emerged from the research namely: Adapting to School; Perceptions of the Neighborhood; Peer Group; Challenges in the Family; Perceptions in the Ethnic Community; and Perceptions of Law Enforcement. These are negative experiences and perceptions which influenced the gang involvement of my participants are discussed.

4.2 - Adapting to school – "They weren't interested in what was going on with me." – Okot

All four refugee young men who participated in this study expressed having challenges adapting to the school setting here in Winnipeg, Manitoba. In some cases, the challenges these young people face at school can be directly linked to their experience as refugees displaced by war. For example, Kanu (2008) and Stewart (2011) state that those

refugee youth who have experienced disrupted schooling due to war have a more difficult time adapting to the new educational system of their host nation. This is because quite often newcomer youth are placed in classes according to their age and not their level of competency. As he reflected on his experience growing up in a warzone and how it affected his schooling, Ermias lamented that:

My only experience is war, the only thing you learn about there is war. [I] couldøve gone to school there and learned good things. Actually, I was a smart kid, but when it comes to school my brother helped me learn to read and write just at home. When I came here they just put me in Grade 8. I thought I was a dumb kid, everyone is speaking English and I donøt know anything, ya know? School was the biggest challengeoschool was the hardest.

It is important to also note that Ermias was 13 when he arrived in Winnipeg and he did not have any prior education. Emmanuel also reflected on his experience on how living in a warzone affected his schooling:

We did school, but as we had to move looking for refuge in different countries there was a huge gap in between that we didnot have a chance to go to school. So when we came here I was 13 and my level of education was grade 4, but the school that we were going to they had to put me in grade 8. My reading level and everything was so poor ya know? So that was a struggle too for me. And never mind the fact that I wasnot really accepted in school ya know? Trying to fit inoit was a hard time.

These experiences can stir up feelings of isolation, depression and of not õfitting inö (Kanu, 2008; Macdonald, 2007; Neumann, 2011; Shakya et al., 2010). Bashir also expressed his experience of trying to adapt to school in Winnipeg:

Even if you sit [in class] they [the teachers and other students] just keep talking and you're just going to be listening to themonothing [not understanding anything]. And some people make fun of you ya know? Like they make fun of you, like your English, they way you talk. It make me feel bad. Sometime I don't go to class if you're pissed off at someone.

Many of the community worker participants also shared their perspectives about the challenges refugee young people face at school. For example, Sadiiq who also came to Canada as a refugee expressed his thoughts about the challenges refugee young people have in adapting to the school system:

If you cannot communicate with anybody, then that disconnects you from the rest of society. It became an obstacle for me to do the things I wanted to do even in sport or at school. If a kid cannot understand what is happening in the class then they are more likely to drop out of school.

Echoing some of the same challenges, Hassan who also originally came to Canada as a refugee shared his experience and views in the following manner:

Language was the biggest obstacle. At age 15 they place you in grade 10. In a lot of classes I couldnot talkóI didnot have the right language to talk in conversations. Even though I really was motivated to do very well and get a high school diploma, it was very challenging. I quit, I quit after 3 years I dropped out of high school. The school system is not working, kids cannot read. Education is the only system that has the kids 6 hours everyday, and what are they doing with it? Kids often feel they donot belong there, they feel dumb, they feel stupid.

Community worker, Jennifer claimed that from her experience working with refugee youth in after-school homework programs, language comprehension coupled with a lack of attention from teachers at school were both key challenges facing refugee youth. Community worker, Saalim acknowledged the improvements that have been made in the school system over the past few years with regards to meeting the needs of newcomer youth however, he also asserted that more work needs to be done:

So you go to this kid, you bring him here they go to school and then they are placed in an age appropriate class. For example, if you are 15 you go to grade 9 for example and you sit in a class and the teacher will tell you read this chapterø, and as a newcomer kid youøre quite embarrassed ya know because youøre just looking at the pictures because you canøt read anything and you find the whole environment strange.

The issues and challenges these young people faced within the school system proved to be too great for them to handle at the time, and all of them were eventually

expelled from school. For example, Bashir shared how his expulsion from school affected him:

When I got kicked out of high school that st the time when I gave up, I gave up already like in life, to go find a job again or go back to high school.

Ermias also reflected on his life shortly after being expelled from school:

I got kicked out all the time, and then I started selling drugs. People say to you fuck school mang they give you money, they were buying me everything, and I was happy right? Because I got what I want. So why do I need school for what I want? I need to follow the love for that moment, ya know what I mean? But I didnot know that in the future, that it affects you. If I did I could we made the right choices. In six years I messed up the whole thing, in six years I could we done a lot of good things.

In addition, Okot shared the following with me about his challenges coping with the school system:

In 2003 that is when it started to get really bad at home and I started getting into a lot of trouble at school, they wanted me to do anger-management, then I started getting a lot of detentions, I got a lot of absents, getting into a lot of fights and hanging around. So they were like this guy is skipping all these classes, he has passed sixteen absences. They tried to work things out with me, but I wasnot complying so they ended up saying well you canot come back to school They ended up kicking me out.

Emmanuel described the events which eventually led to his expulsion from school. This is what he had to say on the issue:

I tried to be strong, but what I was carrying around with me was more deeper thanópeople wanted to fight me, they would say things to me, ya know, but I would just keep it to myself. It was tough, ya know? I felt all this anger inside of me, sometimes I wouldnot go to school because guys want to fight me, ya know? I live downtown and come to school in St. Vital so I didnot really have much friends in St. Vital, ya know? I had so many fears that I couldnot stand up for myself. I didnot want to ruin things for my Mom and brothers and sister so I let these guys pick on me and do whatever they wanted to. Until I was aboutoI think I was fourteen, with all this anger that I had inside of me from trying to deal with all these bulliesoI didnot know what to do. My mind just changed and I was like -whatever, Iom going to school and if anybody do anything to me today thatos it!o

After beating up one of his bullies, he was suspended from school. However, when he returned to school he found he was now accepted by his peers. He reflected on his new found popularity in the following manner:

All these guys that were like bullying me, and people that were staying away from me were now trying to be my friend, ya know, cause now they were like ÷oh man you can stand up for yourself, you can fightø So I felt more welcome, ya know? I felt ÷hey these guys kinda like me, ya know? So thatøs all I had to do was fight, so Iøm gonna do more. So I started getting into so many trouble, and before you know it I started smoking and drinkingóI was fourteen when I started smoking marijuana, ya know? I started skipping school, I didnøt care anymore about anything, ya know?

Eventually Emmanuel was expelled for some of the -activities@he was involved with in school. The expulsion of these youth from school affirms Rossiter and Rossiter@s (2009) assertion that as refugee youth face challenges in school including bullying, they may develop violent responses which in the end only compound their original problems.

Feelings of isolation and of not fitting inø at school led to a change in behavior among my study participants which ultimately resulted in their expulsion from school. The violent reactions of Okot and Emmanuel to their challenges reflect the assertions of Rossiter and Rossiter (2009) and Sersli et al. (2010) who argue that refugee youth may use violence as a way to cope with various stressors in their lives. Rossiter and Rossiter (2009) note that behavior changes such as these among refugee young people are often misunderstood and dismissed as being troublesome. Indeed, Okot reflected this notion when he described some of his trouble at school as follows:

They seen me more as an issue when I started hanging out with the wrong people [there were] a lot of fights at school. They were not trying to tackleóI donæt think they were interested in what was going on with me, they were basically looking at how to prevent this [the fighting] from the school population.

Okotøs story draws attention to how he felt devalued by school authorities, who saw him as an õissueö, a problem. Further, Hardy and Laszloffy (2005) state that õdevaluation occurs when an individual or groupøs dignity and worth are assaulted or denigratedö (p. 25-26). Drawing on the link between a lack of education and gang involvement among refugee youth, Okot noted that despite their lack of formal education these youth still possess valuable life social skills:

If you imagine if you've been in there [a gang] since you were 9 years old and now you've 18 how much school you have missed. These guys can'x read, I know some of my friends who can'x read, like literally can'x read, ya know? But they know how to count money and get the right things.

Community worker, Hassan spoke further about how troubled youth are devalued within the school system:

We dongt value what they already know. Kids that sell drugs have a lot of skills, we often make them dumb, and think they dongt bring anything. How can the school system teach, but also validate their skills and experiences? When schools dongt value these kids, when they dongt value their skills and experiences or care whether they come to class or not, that is the perfect breeding ground for gangs. Thatgs where a lot of these kids met. In gangs they are valued, someone is using their skills.

The statements by both community worker, Hassan and Okot underscore the importance of valuing young people and their various lived experiences regardless if such young people fit into the traditional notion of what it is to be a õgood studentö. Hardy and Laszloffy (2005) have identified ÷devaluationø as one of four aggravating factors that contribute to youth violence.

Polakow-Suranskyøs (2000) research on zero-tolerance policies in American schools sheds light on the negative impact of expulsion from school on youth. The author notes that zero-tolerance policies leave youth feeling isolated and doomed with few options for education elsewhere. In fact, Polakow-Suransky (2000) notes that

õperhaps the most disturbing data trend is the sheer number of expelled students who do not receive an education after being removed from schoolö (p. 107). The educational status of the four young people who participated in this research reflects Polakow-Suranskyøs concerns. Three of the young men never returned to school after their expulsion. The fourth participant only recently received his grade 12 equivalent several years after he was originally expelled from school. This finding underscores MacLeodøs (2009) point, that perhaps schools need to reexamine their role in reinforcing social structures and instead change how their systems operate in order to better meet the needs of these young people.

It is clear that the challenges of inappropriate grade placement and inadequate English language resources left my participants feeling isolated and ostracized from their school peers. Further, their violent reactions and subsequent expulsions from school signal a turning point in their lives in which their violent and anti-social behavior intensified resulting in gang involvement. The fact that only one participant eventually completed high school after they were all expelled highlights the powerful affect expulsion from school can have on a young person.

4.3 - Perceptions of the Neighbourhood – "Why do they put us in this area?" - Okot

All four young people who participated in this study live, or at least were living in the Central Park area of Winnipegs inner city when they first arrived in Canada. As noted in the literature review when refugee and new-comer families arrive in a new country they are often placed in neighborhoods already submerged in poverty and crime. When speaking of immigrant families in the United States Suarez-Orozco and Suarez-

Orozco (2001) noted that õImmigrant families who survive the violence of their countries and the crossing ironically often find a new form of violence as they settle in their American neighborhoodsí All too many immigrant children experience disconcertingly high levels of violence in their new neighborhood and school settingsö (p. 71). Indeed, as in their experience with the educational system, the participants in this study also described negative experiences and identified forms of violence in their neighborhood.

When asked how well he adapted to his new neighborhood when coming to Canada Bashir responded as follows:

Ahónot well cause you have a problem around here [in Central]. First of all people come and fight you, like you canot walk at night time, especially if youore walking alone. I never seen that till I came here. I never got jumped or I never see someone get beat up or somethingóI seen that when I just moved, I started hanging out around here [Central], I started seeing all that.

Further, Ermias identified a direct link between living in the Central Park area and the trouble he found himself getting into:

At the end I blame the government because they put me in Welcome Place. They say it so that so welcomes, they tell you owelcomes, but that so where the gang territory is right? That so where all the evil things happen, so why would you want to put an immigrant that doesnot know anything, to put him in there and then tell them owelcomes.

The government can spend \$20,000-\$30,000 on someone who is locked up like me. Imagine if they spent that money on me, but put me in St. Vital, or Tuxedo when I first came to the country. Put me in the good area, instead of the bad area, ya know what I mean? When you live in a good area, you always learn the good and the bad so you make good choices, you know the consequences. In a good area no one will force you, but if you live in Central you see the same people every day and you're going to have to do something ya know? This guy got shot, this guy got stabbedówhen your friend gets stabbed you have to do something because you're living in this territory where there's gangs, ya know?

Ermiasøtit-for-tat attitude towards the invasive physical violence in his neighborhood and the perception of having no other option but to participate in it, normalizes violence. The normalization of violence occurs as people become desensitized towards it as a result of continued exposure to it (Byrne and Senehi, 2012; Kanu, 2008). Similarly Emmanuel described how easy it was for him to get involved in criminal activity when his family moved to the Central Park area of Winnipeg:

All I had to do was hang out on the streets and sell product, ya know? I felt like that all anyone does, ya know, it kind of feels normal when that all everyone does around you. So that easy, it drew me, ya know became a part of me. When we moved to Kennedy there was guys [selling drugs] all over the place and I had nothing to do so I just hung around. I think if we were to come here and they were to put us in a different community that is nice and not influenced by the drugs and crime, we would we turned out differently.

Contrary to the perspectives of these young men that being settled in a different neighborhood would have protected them from being drawn into crime and gang life, community worker, Heather dismissed that point as follows:

That excuse falls flat on its face umówhen you look at the demographics now because all of the areas [of Winnipeg] have immigrant and youth gang involvement. None of them are safe places at this point if you want to say thereøs no gangs or no negative - thereøs no place like that.

However, a look at the City of Winnipegøs crime statistics website suggests that there may be something to what these young men are saying as the majority of police reported crime in Winnipeg takes place in the Central area where these young men live (oThe crimestat website, o 2013).

It is interesting to note the importance the participants placed on the social environment of their neighborhood with regards to how it influenced their anti-social behavior and negative life choices. All of the young people who participated in this study were settled in the Central area of Winnipeg. Consequently, a comparative study examining the experiences and challenges of young refugees settled in other areas of Winnipeg such as St. Boniface where many French speaking refugees settle may provide

further insight about the role one ighborhood plays in influencing one ighborhood plays in influencing

4.4 - Peer Group – "I never had good friends since I came to this country." – Ermias

All of the young people who participated in this study stated that when they arrived in Winnipeg they befriended other young people who had also previously come to Winnipeg as refugees. This seemed to be a coping strategy for the participants as they noted the people whom they befriended often spoke their native language and had lived in Canada for a longer period of time and thus were able to *show them aroundøthe city. The vulnerability and naiveté of the participants also became a common thread as they spoke about their peer relationships. For example, Ermias reflected on his experience with the friends he met:

Thereøs good friends and thereøs bad friends, I didnøt know these people were bad right? All I know is they speak my language, they know the city good, they know more than me. Some people show you the good things, ya know, and some people show you the bad things, but I didnøt know they were bad people. They donøt tell you itøs going to be badóthey showed me drugs and money, taught me how the drug game goes. They told me they would help me, give me money, buy me a caróand they did, ya know?

Now a community worker, Sadiiq shared a similar experience about finding friends when he first came to Winnipeg as a refugee:

They would give me \$100 just to buy lunch, ya know? They would take me to Footlocker and buy me jeans and shoes, ya know? Buy me MP3, ya know, and then simply tell me that I belong to them, ya know?

The extreme vulnerability, naiveté, and dependence of young refugees when they first come to Winnipeg was underscored by community worker, Heather when she related the story of one young boy on his very first day in Winnipeg:

His cousin who had been here longer and his cousings friend said oletge go to this storeo, so they went to this store and they lifted some stuff, and they told him, oya know you dongt have to pay for everything here in Canada you just walk out with ito and then they said owegre going this way, you go that wayo. And what happened? He got picked for shoplifting. So he was right away on the radar of policeoimmediately, and from there it went downhill.

Okot was lured into the gang life by the attraction of how much money his friends were earning compared to him. He reflected on the day a friend approached him with the opportunity for some oworko as follows:

He was like õhey man why donøt you come drive me around for three hours, Iøll give you \$250.ö And I was like õwhat?! Are you serious?!ö He was like õyeah man I just need you to drive me around for a little bit I gotta go bust a couple of chops.ö So we all jump in my car and I drive them around and I see like wow his phone is always ringing, heøs always talking to people, heøs always making money, and I was like wowí itøs not even like small money, ya know? I think back then people were selling pieces [of crack] for like \$40, ya know, and he was selling like ten or fifteen pieces at a time, times forty thatøs a lot of money, ya know? And I was like õwow, this is seriousö and he was like õyeah man, you should come work for us.ö At the time I was making like \$8.50/hr, I was making like \$500 after taxes and this guy makes like \$500 in two hours! It was just not making sense to me right, to sit at work all day when I have a vehicle and I can make money with these guysóbut it was a choice right? It was a choice for me to go and deal cause it didnøt make sense to me to work for eight hours and make \$500 every two weeks when I can make \$500 in two hours, itøs just a choice.

Emmanueløs experience with his friends also reflects a desire to make money and to be able to provide for himself:

Some were already dealing drugs and making money, having nice shoes, they didnøt have to depend on their parents, theyøll come to school they got money, ya know? For me, my parents, my Mom didnøt give me any allowance cause she didnøt have any. So I was like õman, I want some of thisö so I just hung around with them to see what they do, and they told me everything, ya know?

The stories shared by both Okot and Emmanuel speak to the desire to fulfill one¢s basic needs (see Burton, 1993). These stories also highlight what should be a major area of concern for those who work with refugee youth who are new to Winnipeg. That is the importance of teaching these young people the dangers that are around them, how to

identify these dangers, and also the consequences of engaging in certain activities. The desire to make money reflects a need for skills training and employment opportunities, which can provide these young people with decent jobs and a wage large enough to sustain their needs while also providing an alternative to earning money illegally on the street.

4.5 - Challenges in the family – "Being with the family is still a problem." – Bashir

All four of the young adults who participated in this study identified family conflict as being an important issue in their lives. In-line with what Chekki (2006), Kanu (2008; 2009), Rossiter and Rossiter (2009), Sersli et al. (2010), and Wilkinson (2002) suggest, the conflict in these families seems to stem from the cultural, social, and economic stresses of trying to adapt to new surroundings within a new culture and system of living. In many instances the conflict between parents and children is more pronounced because of the differing speeds at which parents and youth seem to adapt to the new culture.

For example, community worker, Sadiiq reflected upon some of his observations working with refugee families:

The family has a certain way of thinking and certain expectations and if kids donot meet those expectations they are jeered and ya know stigma, and all of this drives kids out of the house and once kids are out of the house, what is next? Youore going to be in the streets, and in the streets what are you going to do? So these are some of the challenges which kids we work with today face.

Similarly, community worker, Saalim shared his views on parent ó child dynamics within refugee families in the following way:

Another thing I find is that when the kids come here they catch up with the culture very quickly ya? And the adults are stuck in the old mindset. It we very difficult

because parents overreact. One parent sent their kid to Calgary to stay with the uncle for smoking weed and now the kid is even worse. They overreact and it drives the kids nuts! The kids say õya know, if you@re preventing me from doing this and you@re pushing me, then I will do it!ö

Bahirøs experience with his parents reflects both the notions of expectations and stigma also noted by community worker, Sadiiq, and he also expresses the same reactionary attitude as observed by community worker, Saalim:

Being with the family is still a problem. When they see you hanging out with someone bad theyøll be like ÷oh youøre badø. And inside you, youøre not bad right? But because youøre hanging out with those guys they think youøre bad and so you get mad. And Iøll be like ok if Iøm bad, even though like you didnøt do it, but people are talking like you did, so Iøll be ÷ok, Iøm going to go and do it thenø Since I dropped from high school everything started changing. As soon as I dropped from high school that means they think youøre bad already.

Bashirøs attitude of taking on negative behavior because others already perceive him to be bad reinforces McCluskey and Torranceøs (2003) argument and Suarez-Orozco and Suarez-Orozco's (2001) notion of *Negative Social Mirroring* that as people search for their identity they may adopt the behavior of the negative stereotypes projected upon them by others. Similarly, Okot noted the following in his story:

If you have no one looking after you and people always telling you you're bad, then maybe you will actually go and do something bad even though you weren't originally a bad kid.

There are also many other family dynamics that have affected the behavior and attitude of these young people. For example, Ermias reflected on conflict within his family and the influences that led him away from living at home:

I never stayed at home since I got to know all those people. At the age of 14 I left home. Those people were feeding me, giving me clothes, money, ya know? My older brother would tell me to come home. Me and him had a bad fight, I was so young, but I tried killing him because he was telling me to come home for curfew. Since that day he just said õlet him do whatever he wantsö, ya know? And I did, and I fucked up.

Cairns (1996) notes that instability within the family can affect the resiliency among waraffected youth.

In addition, Ermias also shared with me how living in a single parent household (his Dad died back in Africa) contributed to his negative behavior:

If my Dad was here, trust me he would lay it down, he would do everything he could to change me, ya know what I mean? My Mom is always busy and sheøs a lady, ya know what I mean? What can she do? Sheøs always scaredóif she says õnoö to me Iøm going to call the cops, or if she beats me Iøm going to call the cops. She doesnøt want her kids to be taken away so she has to give what I want, ya know what I mean? Cause the rules is so much she has to play within the rules right? But my Mom did a lot of things for me man. When I was a kid she actually bowed down to me, ya know what I mean? She bowed down and cried to me õplease donøt do this sonö she cried and cried, and do you know what I did? I walked out when she was cryingóI walked out. That curse has set me back till right now, ya know what I mean? If I wouldøve listened I wouldnøt be in this position right now, ya know what I mean?

Ermias mentions a number of factors within the make-up of his family that negatively affected his behavior. First, he noted that he did not have a father. According to Hyder (2005) õIn some refugee groups it is not uncommon to find that a disproportionate number of households are headed by womení This again has an impact on children, as the emotional impact of living in exile and in isolation, with a lone parent, may well be very difficult to deal withö (p. 35). The observations of community worker, Saalim highlight the implications of growing up in a single-parent family for refugee youth and stresses the importance of a two-parent household for positive development:

Kids who are in gangs, kids who are involved in the justice system are kids whose families have single mothers. Of all the kids I know who have joined the gangs, I dongt know of a kid who has both parents here.

Community worker, Heather who works exclusively with new-comer justice involved young people in her program also noted that othe vast majority were from single parent households.ö

Second, Ermias mentioned how busy his mother always is. Suarez-Orozco and Suarez-Orozco (2001) note that õmigration often reduces the amount of time that parents and children spend together. Many immigrant parents (particularly those coming from poorer backgrounds) work several jobs. These and other obligations make them less availableö (p. 75). Indeed, Emmanuel whose father was killed in his countryøs vicious civil war shared with me what it was like growing up with only his mother around:

It was hard. It was hard man for my Mom, ya know, in the new culture, ya know? At times she was working two jobs just to put food on the plate and I was 13 and I would be the one to take care of my little brothers and sister.

Third, Ermias made a reference to the lack of authority his mother had over him simply because she is a female. Community worker, Saalim also identified this gap as an issue which needs to be addressed within families and their communities:

When you are the eldest boy in the family you assume the role of the father ya? And you¢re the oldest boy in the family, you speak English very quickly, you can¢t listen to your Mom. You assume for yourself the head of the family and you make decisions for the family and it¢s very difficult for you to listen and be led by your Mom.

Further, he noted that:

Here there are different role models, parents have less control. Back home if itøs a single mother it takes a village to raise a child, all the neighbors, all the uncles ya know? Everybody takes care of the kids, but here if youøre a single mother, a poor single mother you have no one else to support you, the only thing is you and your kid who over runs you.

Community worker, Saalim touched on something which Ermias also mentioned when he shared with me his difficulties at home, and the loss of control refugee parents have over

their children when they come to a new country. Moreover, Suarez-Orozco and Suarez-Orozco (2001) note that child disciplining techniques used by immigrant parents which may be acceptable in their country origin may not be acceptable in their new country. The authors state that:

As parents are frustrated and feel increasingly threatened by the encroachment of new cultural values and behaviors among their children, they often attempt to \pm ighten the reinsø But the children, wise to the ways of the new land, may use against their parents the threat of reporting them to state agencies. This further debilitates parental authority (p. 77).

Saalim a community worker, observations regarding discipline within refugee families reflect the above statement of Suarez-Orozco and Suarez-Orozco (2001) as follows:

Kids at school are told that no one can beat you, they are told to call the police. And the single mother who is so scared of losing her kidsóthe kids say õl¢m going to call 911 if you tell me anything, I¢m going to come home anytime.ö And the mother submits to this because of fear.

Community worker, Hassan also shared with me the very same ideas as community worker, Saalim, and added that owe educate the kids, but we dongt do anything with the parents. As evidenced by Ermiasø story of how he threatened to call the police on his mother, community worker, Heather asserted that for many young refugee boys otheyøre adjustment to the system, if you will, is to take total advantage of it.

Yet another issue with regards to family conflict and its impact on the behavior of young people has to do with how well the parents are adapting to the new culture and their new surroundings and circumstances. For example, community worker, Heather noted that although most of the young people in her program came from single parent families, those families which had both parents owere in serious crisis as a family.ö In particular, she noted that:

Fathers have lost their status, the fathers are not adjusting well. For a professional to get into their profession here it very difficult. The father has lost status, who is the son supposed to look up to?

As Suarez-Orozco and Suarez Orozco (2001) also state õparental loss of status in the new society has profound effects on the morale of the parent and hence the childö (p. 76). In fact, Okot identified his father¢s inability to come to terms with his lowered õstatusö in Canada as a key factor in creating problems in the home, as well as being a catalyst for him acting out at school, developing bad behavioral patterns, and ultimately getting kicked out of the family home:

It [settlement in Winnipeg] took a toll more on my Dad than on my Mom because my Mom has never drank or used any substance or anything like that to cope. She was so used to chaos happening all the time, and I started seeing more of that happening at home and then at school I started getting into a lot of fights and things like that cause things were not working at home. Now I can look at it from a different point of view, but back then I didnøt knowóthings at home can affect kids. My Dad was having culture shock because heøs a lawyer by profession and when he came here he expected to be a lawyer and here they were like õno, you have to go back to school.ö So he had to go back and get his mature diploma and that took a long time and in that process he developed alcoholism and it basically brought a lot of problems at home. It was taking a toll on me and my brothers, there was a lot of fights. In 2003 thatøs when it got really bad at home and I started getting into a lot of trouble at school.

Garbarino et al. (1991b) note that the capacity of parents to absorb stress is essential in maintaining õreservoirs of resilienceö in their youth (p. 380). The authors continue by stating õonce that point is exceeded, however, the development of young children deteriorates rapidly and markedly. Reservoirs of resilience become depletedö (p. 380). Indeed, it seems like this was the case in Okotøs family.

Community worker, Heather stated that she has proposed programming aimed at fathers trying to cope with the change of a new life in a new country, and that it has been unsuccessful as fathers are reluctant to share their issues with others.

Okot believes that the families and in particular the parents are the key in social intervention aimed at young people:

The big part of prevention is the families have to work, the communities have to work, and the youth have to work together. Obviously you're gonna need social intervention. The way I see it in Canada, social intervention only happens when there's community intervention and community intervention only happens when there's family intervention. So basically, it's outreach. If you have an outreach program that does reach people then there's going to be a ripple effect. So if families reach out to their kids and say, ya know, owe need helpoo'African families don's reach out, ya know? It's more like we keep what's happening at home, at home. Educate the parents cause the parents are the gateways to their kids.

Community worker, Heather agreed that there is a culture of silence among families with regards to their own issues at home. However, she also noted that overcoming this silence may be easier said than done:

Gossip is a very poisonous force among the immigrant communities. The mothers are terrified that someone is going to gossip about them if they know what is happening. The gossip isnot just about the boy, it is about how she is a bad mother.

The stigma of being a bad mother is so great in some communities that Heather shared with me that she has seen cases where a single mother has began courting a new potential partner in her life, and when the new partner hears from the community that her children are bad because she is a bad mother, he will back out of the courtship.

Overall, Police Officer Smith stated that parents need to be more involved and aware of what their kids are doing:

Many people would be surprised that their children are involved in African Mafia and this sort of thing. I dongt know if they choose to ignore it, if theregs so much post traumatic stress that they dongt even see it happening, or if itgs just that their lifestyle is different. I would never let my 10 year old go out with 17 year old kids. In certain communities everyone all goes together and they all watch each othergs children and thatgs wonderful, thatgs how it worked there, but here if yougher letting your 10 year old go out with a bunch of 17 year olds that kid is going to be involved with drugs.

Perhaps it is true that parents need to be more aware of what their kids are up to, yet in many cases again, this may be easier said than done. This fact is evidenced by the many challenges mentioned above such as: single parenthood, raising several children, children taking advantage of their rights, and parents working several jobs etc. Clearly, there exists a gap where parents and their children are concerned. As some of the participants mentioned, more effort needs to be put into educating the parents. However, the challenge to this as Ermias stated is that õMoms donøt have a lot of time for programs.ö In particular, in cases where there are single mothers they are often much too busy trying to take care of their kids, working one or more jobs, and adapting to a new environment. Consequently, any programming needs to be tailored to their specific needs and challenges.

4.6 - Perceptions in the ethnic community – "They're discarded, they're shunned and discarded." – Community worker, Saalim

Another theme which emerged throughout the interviews was the issue of the gap which exists between youth who have displayed anti-social behavior and members of their ethnic community in particular community elders and leaders. Okot noted how his community turned their backs on him after he was released from prison and had a deportation order against him:

I asked people from my community, õI need your helpö and they were like õnoö. No one showed up [to court]í only my Mom and my little brother. The African way is different than the Canadian way right? Canadians will look after their kids, help them get into a clinic where they can get therapy or whatever, ya know treatment. But the African way is like you@re thrown to the lions, ya know like you gotta survive, it@s do or die, throw you in the water you gotta swim or not.

And they were like õyou survived, thereøs other kids that got killed, do you see that youøre lucky?ö To them itøs survival of the fittest.

Community workers, Hassan and Sadiiq recognized this gap and stated that community leaders need to do a better job of working with these youth. In particular, community worker, Sadiiq noted that:

The community itself needs to be educated so that when a kid makes a mistake and joins a gang, he should not be abandoned by the whole community. Like if you're a Somali today and you're been in jail the whole community will be saying olook at this criminal kido ya know? So kids will never feel comfortable you know what I mean? Once you're bad, you're bad.

Moreover, community worker, Heather stated that when her organization went to the leaders of one of the African communities (coincidently the same community that Okot, and community elder, Abraham hail from) for help in knowing how to address some of the issues with these young people, they only responded by saying that the organization is doing the best that it can and it should keep doing what it is doing.

Community worker, Sadiiq also noted that there is very little in the way of community run or church / mosque run initiatives to reach out to these youth. Where there are ethnically based community centres or programs Okot stated that in his community these organizations are not reaching the youth:

The community has a lot of resources, ya know? They get a lot of grants from the government, they dongt invest in that cause they dongt see the youth in there, ya know? They talk about the youth, they go to the government and say owe need to help our youtho, but when they come to the community centre Igwe never seen a youth going in there and hanging out with the older people, it is always older people in there. The only time yougd ever see a kid in there is when it is a community event, and even then they stigmatize, they gre stuck all the way in the back and all the older people are in the front. When they gre talking to the crowd, it is talking to the older people not the young people.

They dongt have any programs to help these young people get out of gangs, or even address the issues of gangs and the lack of education they have. The amount of time Igwe spent talking to you right now, I cannot do that with any of the

leaders in my community. They dongt have that much time to spend with me, ya know? So if you dongt have enough time to spend with me right now and you say the youth are the future? If I have a chance to expose this community I will, we need to address this!

He continued by stating that the community centre should be open more than just 2-3 times a week, and should include youth programs, and at the very least should be a place where youth can go and grab a sandwich or tea instead of hanging out on the street. The perception of the lack of community support from one own ethnic group was not just that of Okot, it was also the opinion of Emmanuel who is from a different ethnic group and completely different part of Africa:

In the African community the leaders areóI dunno I donøt think theyøre veryóI donøt want to use the word æducatedøtowards this kind of issue, but at the same time theyøre kind of afraid, ya know? They donøt have the boldness to go after youth. They will sit in their offices and only when the kids come in then they will be able to approach them, but when they see the kids out there, ya know, they donøt wanna go talk to them. They donøt care, ya know, theyøre just like õlet them beö. And some people look at them kids and they donøt even know what theyøre going through or even though they know a kid is dealing drugs, ya know, you know what heøs doing, they donøt know why heøs doing it. You donøt know why heøs hanging around these people, ya know? Maybe sometimes he just needs a place to stay away from these people, ya know? He donøt feel welcome. Most youth just look at these help centers as just a joke, ya know? They donøt get down to the root problem, ya know? Like the cause of why these kids do these things.

When asked about the role of ethnic communities and cultural activities in reaching out to young people, the words of community worker, Hassan reflect some of the challenges and difficulties these communities face:

People that worry about culture are people who already have their basic needs met. When people come here they dongt have their basic needs. Itsø hard man seriously. Youghe trying to go to school, youghe trying to adjust to a new environment, youghe trying to work a couple of jobs, youghe trying to do so many things. You dongt have time to adjust and celebrate. We do not have enough culture, it would be great but we dongt.

Abraham, an elder from the same ethnic community as Okot acknowledged some of the struggles young people in his community face such as: stress, trauma, financial struggles, and not fitting in at school. However, he also asserted that the community is trying to reach out to young people, but reiterated some of the same challenges they have in doing so as community worker, Hassan:

People in the community are very busy trying to put together their own lives ó money, family, struggling to find a job. There are no retirees who could do this [run youth programs] because the community who is here is still quite young. All of us are students with families and dongt have a lot of time.

Further, community elder, Abraham noted that his community would like to solve its own problems, yet it does not receive the funding despite its efforts to submit many funding applications to local government. He noted the need for more support on the family level as well as for community based initiatives.

The findings from the discussions regarding the ethnic community presented a wide gap in perceptions by the refugee young people when compared to those of community elder, Abraham and community worker, Hassan. There is a clear lack of communication between young people and the leaders of their ethnic communities. In particular, Okot perceived his community as having a lot of resources, which simply did not trickle down to the young people, and who also did not care enough to make time for their young people. In contrast, Abraham an elder from the same ethnic group as Okot had a completely opposite perception stating that the community centre actually does not receive very much funding and that it simply does not have the resources to assist the youth. In addition, he stated that people in the community are themselves busy trying to put their lives together and as a result many do not have the time to run youth programs. In order for newcomers to build cross-generational understanding these two different

perceptions of truth need to be reconciled and this can only happen if the young people engage with their elders or if the elders engage with their young people. Either way, someone needs to make the first move.

Strengthening and building support for ethnic community organizations helps to build social capital for newcomer families and õis considered to be a key factor in the immigrant adaption processö (Chekki, 2006, p. 4). Ethnic and cultural organizations not only provide new-comer families with ethno-cultural support already in place, they also provide newcomers with a sense of social and spiritual fulfillment, and a sense of identity, belonging, and pride (Chekki, 2006). In the case of my participants this key support system is clearly not reaching the young people in the community.

4.7 – Perceptions of Law Enforcement 6 "One of the biggest challenges they face now more than ever is discrimination. Like the police discrimination ya know?" – Community worker, Sadiiq

Critical perceptions of the Winnipeg Police Service were expressed by both the young adult participants, as well as the community worker participants. The interviews unveiled a gap in communication and trust between police officers and refugee young people. Ermias pointed to the police as an obstacle for him changing his life into something positive:

The police, when they pull you over you get a fake charge. Because we were associated or related to [name of gang] before, so they think we're still doing that, but what they don't know is we're trying to change man. So who's trying to do the same thing over and over and over? That's just dumb. If they would see me right now, if they'd pull me over they would just give me a fake chargeothey would just make it up! And the judge will believe them because I'we been associated with [name of gang]. Even if you beat the case, it will still be under your name on your record. The police fucked up my life man, that's why I hate

this country. I dongt want to blame my problems, but they just add it more instead of helping me.

With the youth that he works with, community worker, Hassan has also noticed a similar experience to that shared by Ermias:

When the kids are in the system the police will harass them more and more, and the cops are looking for a reaction from them, to get a charge against them for assault or something. For sure there is racism, for sure there is ignorance, they want them to stay in the system.

Community worker, Saalim also noted the frustration some young people feel from their interactions with police:

Thereøs a lot of racism by the Winnipeg Police Service towards newcomers. The frontline officers are not quite educated on cross-cultural understanding. The police see these kids in groups wearing hoods etc. they think they are doing deals, there is constant harassment. They push them against the wall five times, and the kid is going to be like õwhat the heck man Iøm going to do it, they hate me so what? Iøm going to prove it to them that I can do it.ö

The frustration noted by community worker, Saalim reflects the notion of *Negative Social Mirroring*. Community worker, Sadiiq noted that the way many African young people in Winnipeg dress ó wearing hoods and baggy pants is a reflection of what they see in society and does not mean that they are involved in gangs:

We are dealing with an identity crisis with these kids. They are no longer in Africa, and they cannot relate to white people. The only people they can relate to are other blacks, and the other blacks happen to dress in a certain way, so for them it a statement. They are saying ya know ol belong to some people, I can relate too.ö

Okot and Emmanuel claimed that they have been taunted by the police. Okot stated that a police officer once told him oby the time Iøm done with you, youøll be lucky if you get to stay in Canada.ö In addition, Emmanueløs experience with the police also reflected a macho and arrogant tone as he was once told by a police officer that owe are the gang, we own these streets.ö In other instances community worker, Sadiiq noted that

õpolice do not talk nice to them, they use the \Re word, sometimes this could be a black police officer, it doesnot matter, itos just police in general ya know?ö Negative and confrontational attitudes as perceived by these participants reinforces the gap that exists between police officers and refugee young people.

Police Officer Smith also shared a perspective on the gap that exists between the newcomer community and the police:

We know thereøs social issues, but we almost go we need to get tougherø. People donøt want to fail, our newcomers arenøt moving here wanting to fail, theyøre not coming and saying õl want to join a gang, thatøs what I want to do.ö So we know thereøs social issues and we think thereøs huge mistrust with the police and we need to be in my opinion training our communities to take ownership of their community again. Iøve seen many newcomer leaders do this, they say õweøre really supportive of the police and we really want to make a difference and build that relationship and weøll do that when you guys stop pulling over our kids.ö And itøs like, we pull over a lot of kids, itøs our job to watch for things. Many of the parents whether youøre a newcomer or Canadian born and raised, parents just donøt believe the police anymore, they donøt want to believe their kid is a gang member. They donøt want to believe that ya, sometimes weøve seen your kid do a couple of things and weøve picked them up a couple of times and you still donøt believe me.

Further, Officer Smith stated that:

I think what we're lacking is that the community needs to know how to police itself. I don't mean like making a citizen's arrest, but just being aware of what your kids are doing. And trusting us that if I say your kid is in African Mafia or if I say your kid is running drugs, trust me. There's a gap in our justice system, yeah it's great that you have programs in the youth centre, or in Stony, or in Headingly, but are you getting to the issue of what's going on? They're making headway right now in our Aboriginal population by doing just that, by talking about experiences with residential schools and the trickle-down effect from parents. And we need to be doing that with our newcomers, but we don't know enough about it so now the elders of the community need to step forward. The community needs to back the police and the police need to back the community, otherwise it's never going to work.

Community worker, Saalim agrees that the whole approach to dealing with some of these young people needs to be changed:

The thing that I find which is very difficult for younger kidsóis ya know, the justice system. You arrest a kid who is 13 years old, 14 years old who was found selling a joint one day. Maybe that his first day, maybe that her first day and you take them to the youth centre where they spend two weeks and they come out graduates. When you come out you are hardcore, you know the deals, you know the language, you know how to do it. And you become full-fledged, nobody has any control over you. Instead of like ó ok this kid was found selling drugs for the first time, does he have a record? Ok he doesnot have a record, how do we deal with this kid instead of taking him to the justice system ya know? What the best way to approach?

Community elder, Abraham noted that in his view the police were doing a solid job, but the gap that exists is between youth based organizations and the ethnic communities. He stated the following in his story:

There are programs I don¢t like, like [name of organization]. Kids that have enrolled with them I haven¢t seen a child come back rehabilitated. What you are doing is not working, so stop it!! There needs to be more communication between organizations and the community. Kids often fall in and out, there is a need for sustainable growth. You can treat malaria, but if you¢re sleeping without a mosquito net then the problem will return.

In order to address the gaps which exist between the police 6 newcomer communities 6 and refugee youth service organizations, an initiative termed *Umoja* meaning aunityø in Kiswahili has been formed by leaders in the refugee community along with the Winnipeg Police Service to start a dialogue between these three groups. According to the Winnipeg Police Service officer participant at times it ôhas been a struggle to get people from the community to come.ö However, despite these growing pains *Umoja* has taken some positive steps. For example, community worker, Sadiiq stated that he now does presentations to cadets and new recruits in order to help them gain a better understanding of the various challenges and perspectives of refugee youth in Winnipeg. Further, Officer Smith stated that cadets are now participating in after-school homework and sports programs for refugee youth in order to build trust and

understanding on all sides. For community worker, Saalim this type of relationship building is something he would like to see more of in the local community:

Isnøt it better for a police to play with the kids, to play basketball with kids so the kids trust him instead of trusting a gang member? So the role model, the person who the kids trust and look up to is the policeman rather than someone who is on the street. So why canøt we do this? The Mayor doesnøt get this.

Saalim also noted the Police Citizens Youth Club (PCYC) program model used in Australia where police are heavily involved in running youth programs as something which should be tried out here in Winnipeg. Officer Smith agreed that this type of approach would be a great idea, but there simply just are not enough resources available to create such a program.

Lack of resources is another gap that exists in the area of service provision. Community worker, Jennifer stressed the importance of resources in providing refugee young people with the essential support needed to avoid falling into gangs. However, community-based organizations as well as agencies such as the Winnipeg Police Service can only do so much as their funding allots for them to do. Quite often the bulk of such funding comes from at least one of the various levels of government. However, as community worker, Heather noted:

The government wants to portray itself as a welcoming place for refugees without giving much forethought on what it is like to come from a refugee camp.

Therefore, if agencies, communities, and cities are serious about addressing these issues they must petition the various levels of government to provide more funding to these areas or search for alternative funders. With such a holistic, multi-modal and multi-level approach also comes the responsibility to ensure effective, efficient, and appropriate programming.

4.8 - Key Findings of the Research

There were several significant findings, which emerged throughout the course of the research. These findings were indentified within the various themes of Chapter Four.

Based on the interviews with the participants it was clear that they all had a difficult time adapting to the school system in Winnipeg. The reasons for this difficulty stemmed from grade placement based on age rather than competency, and an inability to communicate with others due to language barriers. These challenges left my participants feeling isolated and made them vulnerable to bullying and gang recruitment. Further, it is clear that the schools were not equipped to meet the unique needs and challenges my study participants faced. These negative experiences at school led these young people to associate with other likewise disenfranchised young people with whom they could relate.

The fact that all these young people were expelled from school for their behavior points to a gap in the school system whereby teachers, schools, and the school divisions are simply not equipped to identify and address the root causes of their negative behavior. The fact that only one of my study participants eventually completed high school many years after being expelled highlights the significant affect expulsion from school can have on a young personose life chances. As well, this fact also suggests that if schools are to truly be a democratic and inclusive place of learning, growth and nurturing for all students, we need to consider other ways of holding them accountable for their actions, while also identifying and addressing the root causes of their negative behavior.

It may be short sighted however, to suggest that expulsion from school alone is the determining factor in a young person continuing down a path of anti-social behavior. All of my participants also lacked essential supports in several other areas. As noted both in the literature and in the interviews with my participants, refugee young people adapt to their new environment much more rapidly than their parents, which has created conflict and power struggles within the family home.

Through schools refugee young people are made aware of their rights as youth living here in Canada, rights that they may not have been able to exercise in their home countries. Their awareness of the laws has become a particular issue with regards to how parents discipline their children and the authority they hold over them. As evidenced in the literature and interviews, young people are using the threat of calling the police and Child and Family Services (CFS) when threatened with discipline from their parents. The fear of losing their children to the authorities has left parents feeling helpless so that they allow their children to do what they want. The abuse of knowledge on the part of young people coupled with the fear and loss of control the parents are now experiencing identifies a gap in which both refugee parents and young people need to have access to resources, which can inform them of alternative ways to solve conflict within the family.

Further, as noted in the research many of these families are single parent households often headed by the Mom. In many cases mothers do not have a sufficient grasp of the English language. They are also raising several children while working, often more than one job. In short, many mothers are not equipped to nor do they have the time to assist their kids with school homework. This lack of support at home is yet another factor in refugee youth not adapting well to the school system here in Winnipeg and may contribute therefore, to their negative behavior.

The importance of a strong father figure in the lives of refugee young people is another key finding of this research. Both Ermias and community worker Saalim noted

that young male refugees may have difficulty accepting the authority of their mother as the head of the household in the absence of their father. In refugee families who do have fathers present, the fathers often experience a drop in social status as their professional credentials in their country of origin are often not recognized in Canada. Community worker, Heather noted that this loss of status makes it very difficult for young males to look up to their fathers as key mentors in their lives. Further, Okotos experience shows the demoralizing and devastating affect the loss of status can have on a family as his father developed alcoholism as a result of his culture shock. The affect of Okotos fatheros alcoholism resulted in violence in the family home, which ultimately led to Okotos increased anti-social behavior and gang involvement.

Conflicts within the home whether it is a from a son refusing to accept the authority of his single mother, or the family dysfunction caused by an alcoholic father trying to cope with the loss of social status, are further compounded by the culture of silence and fear of being the victim of gossip noted by both community worker, Heather and Okot. This finding underscores the importance of creating support programs aimed at single mothers, as well as fathers coping with the loss of social status that encourage them to share their experiences and struggles so that they do not suffer through these family conflicts in isolation. However, arranging such programs may be difficult as my participants stated that single mothers do not have the spare time to attend these programs. As well, getting fathers already struggling with their identity and loss of social status to open up to one another about their issues and struggles may also prove to be difficult as noted by community worker, Heather.

The ethnic communities of my participants fell short of providing the necessary support mechanisms needed to ensure the resiliency of these young people. The lack of support from the ethnic community stemmed from three areas. First, as noted in the interviews many potential adult role models in the community including the leaders and the elders are themselves busy with their own families, struggling to make ends meet to secure a future for themselves and their families. Thus, they do not have the time or energy to support these young people when they themselves are also looking for support.

Second, there appears to be a lack of understanding of how to approach and address the negative behavior of these young people. Possibly, this lack of understanding exists because anti-social behavior such as criminal activity, drug use and gang involvement may be new problems and not previously experienced by the community. Moreover, current methods of dealing with these issues are simply not working in this new context, and it is very clear that shunning and/or discarding these young people is not working as a method to correct their behavior. Communities and in particular their leaders and elders need to work with the families and the young people in order to address the root causes of their negative behavior and to prevent gang involvement.

Third, there is a clear lack of communication between refugee young people and the elders in their communities. Community workers, Sadiiq, Saalim, and Hassan all stated that in terms of the discipline and mentoring of young refugee males, the community leaders and the families need to communicate better with their youth and not discard them once they have made a mistake.

Okotos perception of the elders and leaders in his ethnic community and the work that they do, or do not do for the young people in their community was at odds with what

community elder, Abraham articulated, and also points to a lack of communication between community leaders and the young people. Okot perceived the leaders in his community to be receiving a lot of funding, which was not trickling down to the young people. Abraham on the other hand, stated that the leaders in the community would like to run more programs for the youth, and in fact do not receive adequate funding to do so.

Unable to find the supports they needed at school, in their families, or from their ethnic communities, the participants for this study had no problem finding alternative systems of support in their neighborhoods through street level criminal activity and gang involvement. The importance of geographical location when newcomers are settled in Winnipeg was not lost on my participants. All of the participants recognized that their inner city neighborhood in central Winnipeg is overrun by crime, violence and gangs. This reality made these vulnerable young people easy prey for gang recruitment. My study participants noted that the friends they made from their neighborhood were negative influences on their lives and led them into gang involvement.

Many of the participants thought their lives would have turned out much better if they were settled in more middle class and affluent neighborhoods, where they perceived that the negative influences of crime, drugs, and gangs do not exist. Moreover, the issue of negative friends and the fact that they are preyed upon by gangs underscores the vulnerability of refugee young people when they first arrive in Winnipeg. This vulnerability also highlights the importance of creating early prevention and awareness programs as some of my participants have suggested so that when young refugees arrive in Winnipeg they are aware of the various dangers within their new neighborhoods.

Newly arrived young refugees are particularly vulnerable in cases where they have cousins who have been in Winnipeg longer and who are already gang entrenched.

Another key finding of this study is the negative perception of law enforcement by my participants. Experiences with racism and police harassment have led to a deep mistrust between refugee young people and law enforcement officers. On the other side of this issue it appears that the police would like to see more engagement from community leaders and parents in õpolicingö their own communities. This gap is being addressed by the *Umoja* initiative, but again, this is a slow process which requires commitment from all sides. The police need to be committed, and parents and community leaders need to take time out of their busy days to attend these meetings. Including educators in these gatherings would also be a positive step towards building a stronger and more understanding community. However, as noted in my interviews this has sometimes been easier said than done. In order to move forward and close the gap, which exists between law enforcement and newcomer communities a solution that meets the needs and fits within the schedules of all concerned parties needs to be discussed.

The reoccurring theme of physical violence experienced by my young male participants in each of the six theme areas presented in this chapter is another key finding. My young male participants discussed witnessing violence in their countries of origin, in their neighborhoods and in their homes in Winnipeg. They also discussed being victims of violence in their homes, in their neighborhoods and at school in Winnipeg. Finally, they spoke about their experiences with participating in violence in their homes, at school and in their neighborhoods in Winnipeg.

The culture of violence these young men grew up in has taught them that violence is normal, and in fact an appropriate way to address problems in their lives. The different areas where these young men experienced violence are interrelated as the experience of Okot provides a good example of this. When Okot experienced violence at home, it led him to respond with violence against other students at school. His violent acts at school ultimately got him expelled. He then found himself in the streets with his gang where he was both a victim and a perpetrator of violence in his neighborhood. Identifying the interconnectedness between these areas where violence occurs is essential to addressing structural violence and getting at the root causes of why a person such as Okot would participate in anti-social behavior. As well, the interconnected areas where violence occurred with my participants mirrors the multiple areas of risk noted by my respondents such as: school, family, neighborhood, and peer group.

4.9 - Conclusion

Chapter four has presented information gathered during semi-structured interviews with the study participants. In so doing, six key themes emerged in which my respondents noted how the negative experiences and perceptions have contributed to their gang involvement. In particular, areas where support was lacking were identified as school, family, ethnic community, law enforcement. Where positive support systems were lacking, negative support systems based on criminal activity and gang involvement accessed within the neighborhood and through peers attracted these young men. The interviews with these young men, the community workers, the elder, and the police

officer make it clear that more needs to be done in terms of providing necessary supports within schools, families, neighborhoods and ethnic communities.

Chapter 5 – Walking Away from Gang Life

5.1 – Introduction

Chapter five focuses on key themes that emerged during the interview process, which highlight my participantsø positive experiences and discusses the pathway out of gang life. This chapter discusses four interviewee themes including: Influences and Role Models; Turning Points; Coping Strategies; and Addressing the Issues. My participants shared their positive experiences and perceptions as well as discussed their difficulty in leaving the gang. Further, my participants shared their points of view about how to address some of the challenges they have dealt with so that young people similar to them do not have to go through the same struggles.

5.2 – Influences and Role Models – "You're a blind person when you don't know about life." – Ermias

Despite the many negative experiences and influences as noted in my study participantsø various interactions with school, family, friends, ethnic community and neighborhood, the young people in this study also noted sources of positive influence.

The following statements shared by both Bashir and Okot highlight the positive affect a caring and thoughtful teacher/mentor can have in empowering and influencing young people. With regards to how his shopsø class teacher helped him, Bashir shared the following with me:

I thought that I canot do it, like I failed already. Then I made a belt buckle, candle holder, welding, I did all that. Then I think, õI can do it, itos easy!ö

This positive experience sparked an interest in Bashir as he stated that one day he would like to have a profession in mechanics or welding. It is doubtful he would have this same

interest today had his teacher not given him the time and attention that he needed in order to succeed. Similarly, Okot shared with me the impact a soccer coach had on his life as follows:

My teacher did actually try and reach out to me, and they put me in this after-school soccer program. And that worked for a bit cause I ended up winning two championships back-to-back. That was in 2001 ó 2002, but in 2003 things started getting really bad at home and I started getting into a lot of trouble.

Although his success in soccer was not enough to quell the storm of his family problems, knowing that his involvement in the soccer program provided at least some positive experience is important to note. The role of positive mentors and role models is something teachers and community workers can use to build on in providing these young people with more constructive experiences in their lives.

Community worker, Saalim indentified school sports as a key area to empower youth who are struggling in school:

These kids are very good at soccer when they come here. So instead of using soccer to empower them, you introduce them to basketball, and hockey, and football which they dongt know and they suck at it, and they find ya know \pm canget do anything@and they become so helpless. So instead of doing that, why canget the schools use soccer to empower these kids? When they are able to do soccer and they do very well, they feel empowered and can easily be introduced to other things.

The importance of having an engaged teacher who is interested in supporting a young personos interests and in encouraging his/her talents in order to build success was also noted by community worker, Sadiiq who shared his experience as follows:

There were days when I would say õfuck school, Iøm going to find a jobö, but finding a job was hard and it made it easier for the gangs. They would say õnobody is employing you, you donøt speak the language, street language is easier to learn you donøt need grammar, no school needed, youøll be good at it, youøre a tough guy, youøre a clever guy just do this little bit for us every day.ö Then I would say õfine Iøll do itö, but then the next day I would say õno, I cannot do itö, and then I would say õyes, I can do it, no I cannot.ö

Then I became friends with one of the teachers at school and he kind of helped me. I dongt think he understands what he was doing, the impact of what he has done. He connected my childhood and where I am now. He emphasized that I should be proud of who I am and look forward, and move forward, ya know? So that made me realize that I cangt go this road [gang life], this road does not reflect my challenges. Then after that I started thinking what I can do, ya know, rather than just looking after myself. Then I started doing documentaries about the challenges of peopleí Just out of that conversation came finding out what yougre good at, ya know? And then I knew that I am good at writing stuff, so I started writing and once I wrote this stuff then I talked to the teacher about it and he thought that was an amazing project. And I said I can make a film, and he said yes and then that helped me to stay out of gangs literally, ya know? I was able to kind of manipulate what was happening in the gang life into the film, ya know?

The experiences stated above highlight the great influence and positive affect a teacher can have on a young vulnerable person go life. As noted earlier, schools have our young people for five days a week and six hours a day. Within the time spent at school rests an enormous responsibility and an equally great opportunity to engage these young people constructively in a, caring, and interested way.

Community organizations that run youth programs also have a great opportunity to reach out to vulnerable refugee young people. For example, Bashir explained how one organization program helped to wean him away from the gang culture:

[That program] helped me. Volunteering with the kids in the hip-hop program doing music, playing soccer and basketball. Hip-hop makes me feel fun, doing performances for the kids, in rapping you dongt have to talk about gangs or anything, that what I like about it. It fun, you write your stuff and perform it, it just makes me feel happy. Sometimes I write about my friends, I talk about Africa, the war in Africa.

However, Bashir noted that since returning from a stint in Alberta he has not been motivated to rejoin the program.

Ermias also shared his experience of being in a program specifically aimed at refugee youth at-risk of joining street gangs:

I would go that program, they chose me cause they knew I was going in a bad way. That program told us a lot. They would show us videos of how gang members get killed or deported, and everything that was happening in that video is happening to me right now. As I speak tomorrow is my court date and I might get deported. What are you gonna do once a week? The programs should be everyday, cause on the outside [in the street] Iøm learning a lot. When you donøt have a lot of people in your life to help you itøs sad man. My life got corrupted for me not knowing.

The infrequency of available programming that Ermias refers to for vulnerable young refugees such as himself is very significant in that, when the pull of the street is so strong, it seems unlikely that a oonce a weeko program will have a lasting effect.

Although the positive experiences shared above did not ultimately prevent the participants from joining a gang, they nonetheless provided a glimmer of hope of what could be achieved with more case appropriate and targeted resources that bring together family, school and community. However, according to community worker, Jennifer, there simply are not enough resources in place to address all the issues these young people face. Again, this is most likely related to a lack of funding and positive role model volunteers for such programs.

Okot spoke about just how major an influence life in the street can be in the absence of effective and available programming:

Theyøre just so used to like having nothing, Itøs not a surprise that they want to hang on to the only thing that they have. Thatøs what the government doesnøt understand. Oh these kids are always bad, these kids are this etc.øBut in order for you to get these kids off the street, which is the only thing they have to survive, the only thing theyøre good, at ya know? Theyøve solidified themselves, I have a future in this, they see that. Iøm a boss now, I got money, I got a life, I got a house, I got my kids taken care of, ya know? Like they donøt think like of how you guys would think of them. Theyøre thinking of securing their future, ya know? Iøm gonna get \$250,000 out of this and start a business, keep this thing going on the side, start a music label, whatever. Like they want a way out, ya know? But thereøs no way out to them, thereøs nobody reaching out. So the only thing is going to be like well nobody reached out to me what am I going to do? Iøm going to keep doing the same thing. Iøm making money this way for sure, I

wake up every day my phone is ringing, I got people bringing me money, why would I stop that? What do you got for me to change? All you want me to do is do you, do what you say, but you don¢t have a substitute for me. That¢s what I found really hard for me when I was changing. I didn¢t have a way out, all I heard was people telling me õyou¢re a bad kid, you need to get out of this, you¢re messed up,ö but they didn¢t tell me how.

Similarly, Bashir also spoke about the difficulties in leaving the street life behind and the need to find an alternative way of living:

When I sit by myself Iøm like õoh I need to go to school, find a job, make myself leave from hereö, ya know, try and move on. But every time you try to do that thereøs something that blocks you, ya know? Something stops youóI donøt know what to say about this, cause I have a lot of struggles. Sometimes I think õwhat am I doing?ö I never used to be like this so I try and move on. But you know, you donøt know where to start from right? But you need to find something to move on with, if you donøt find something to move on, youøre not going to move on.

Reflecting on what can be done to motivate youth to leave gangs community worker, Sadiiq noted the following:

I think employment is number one. If we have more employment for these kids they would be more than happy to come clean and work. But when a kid makes a mistake and he gets a criminal recordóforget it! How can he find a job with a criminal record? So once he makes that mistake theyere like offorget it, Iem not coming back!ö

Echoing the sentiment that employment opportunities are needed, and effective programming is essential in saving a young personøs life, Ermias spoke about a job placement program he was once involved in. This is what he had to say:

Those people told me they would get me a job and help me. And they got me a job and I worked a full two years, I was working straight. I forgot about everything, ya know? I was doing good, going to school, going to work. That program helped me cause they got me a job. Then the program got cut, everything got cut, my life went down. I started putting myself down. I was doing good for 2 years, I was trying to go to work, I was trying to go to school and in the end I ended up losing everything, ya know? After that I stayed at home for a year and a half doing nothing and my Mom was always telling me õgo get a job, go get a jobö ya know what I mean? And then the same people I met on my first day came to me and said õletos go do thisö and I did, I made the wrong

choice. I went back to selling drugs for another 2 years and I got caught and now Iøm facing the consequences. Now itøs like they want to deport me. Theyøre trying to give me 22 months and then Iød have to do immigration time. I canøt keep saying õI made a mistakeö but I donøt know what else to say to them. If my Mom found out I was getting deported she would have a heart attack, ya know what I mean? If that program wouldøve continued until today I would have been myself, ya know what I mean? But for some reason they cut the program and I donøt know why. When they cut the program, they cut my life too, they cut a lot of peopleøs lives, a lot of people got locked up after that.

However, even in spite of available programming and a desire to leave the gang life behind, Okot explained that it can often be easier said than done:

Some get recruited when they are really young, like those who I told you about who got into the game when they were 9 and now they re like 18, 19. Some bring their younger cousins and those kids have a harder time getting out cause the game works like if I brought you in then you re loyal to me, ya know? 6 to the end. These kids will always be loyal cause like when I had nothing you were there for me, ya know? I had nothing, I didnot have a home, I didnot have food, you gave me like all these nice clothes all these nice shoes, you gave me a place to sleep, you gave me money in my pocket, ya know? You showed me how to get money. I was staying with you, ya know? So you gave me all these things when I didnot have anything, and then when I got to a point where I could sustain myself all you wanted me to do was continue to push for you, ya know? To stay connected to you, stay loyal. So how you gonna break that, ya know?

What he is raising here is a desire to belong, to have someone look after him, and to have his basic needs met. If nine year old children feel more looked after in a gang than they do with their own family or community then we as a society have failed them.

Okot shared his own story with me highlighting the difficulties of leaving his gang:

I had a weapon I would keep around. They literally tried to kill me a couple of times. They came to my house like five or six guys and they tried to likeóthey did try like two or three times at least to try and really get rid of me. They were like paranoid right? Like this guy is up there, he knows stuff, he might try and snitch on us. But that on the point, I on trying to do this for me, I on trying to clean myself up right? But to them, they are like ono we do not need to lose another member, we need you to stay of they of try and force me to stay. But there os no way around it, I can hide. I realized that if I want to get things accomplished I have to go out, and whatever happens, happens. I went from carrying a weapon to

like I put it away, I buried it. I was like forget this, Iøm not going to walk around scared. Iøm going to show myself that Iøm not afraid and Iøm going to show them that Iøm not afraid. For these guys they go by respect right? They jumped me a couple of times, but after a while that all stopped.

Emmanuel also noted the strong expectations of remaining loyal as even after he moved away from Winnipeg to a rural town to get away from gang life, the gang threatened him with violence if he did not continue to purchase drugs from them to sell. As a result of this threat Emmanuel was lured back into the gang where he continued to sell drugs for them in his new town until he was eventually arrested and sent to prison. Emmanuel reflected on this experience in the following way:

At that point though I had been through so much, like being out here [in the rural town] and seeing how things were going. I was kind of pulling myself away from it slowly, but it wasnot working. I didnot know anything to do man.

These experiences signal the need for effective supports that empower young people to leave gangs in a manner which protects their personal safety. In some cases this may mean confidential relocation to a safe house with the necessary resources to ensure a successful transition for that young person.

There is a battle for the lives of these young people between the lure of gang life and the positive influences of mentors and appropriate programming and services. Targeted programming, which is relevant to the lives of these young people can work if it is persistent and sustained. It is quite clear that the pull of life on the streets is powerful, and thus the frequency of available mentors and accessible programming needs to increase and be sustained in order for these young people to make positive and healthy lifestyle choices. In particular, schools and community organizations have an important role to play in nurturing a culture of belonging, positivity, and healthy relationships.

5.3 – **Turning Points** – "Going to jail was a real eye opening moment for me." – Emmanuel

All of the young adult participants shared with me their ah-a moments in turning their lives around. Not surprisingly, all of them noted a significant life event, or a series of significant events that ultimately led to their decision to leave gang life and gang association behind. For example, Bashir was reluctant to go into detail with regards to the impetus that influenced him wanting to change his life, but he did make reference to a violent incident in his neighborhood, which affected him deeply. For Ermias it was the threat of being deported unless he cleaned up his act that became his wake up call. Further, Okot also noted that it was a series of life events that influenced his decision to turn his life around:

The sheer reason for me to change was likeóI came to terms with myself. I understood that when I got out of prison things were way different, it was a larger group, there were weapons, there were guns, bigger number of people and you can get killed over something and I realized I didnøt want to be a part of that. For me to stay on top I would have to be more ruthless, and I would have to basically erase some people and I didnøt want to be involved with that. I knew these guys were serious cause some people lost friends, like people got killed. I realized that in order to stay on topóitøs not that I canøt, itøs just that I chose not toólike I had people that followed me and said that õI can do this for youö, ya know? I still knew people who could connect me with a large amount of coke, extacy, ya know, heroine whatever I need, ya know, weapons, it was not hard. I remember I used to buy like weapons for like \$75 for protectioní like kids would just have hand guns like it was no problem. I realized that if I wanted to stay like that, I would have to become more like that, and I knew I wasnøt going to be like that.

Although he maintained a connection to his peers in the gang and still struggled with addiction, Okot slowly began to turn his life around by gaining legitimate employment, returning to high school, and seeking addictions treatment and therapy where he credits õa really good counselorö for helping him move through some of his issues. However, in 2009 a friend of his was murdered as a result of gang violence, and it was at this friendøs

funeral where he finally realized he needed to end his relationship with the other guys in the gang:

In the eulogy one of my elders he was saying õyou guys need to stop fighting. At home we came from a war country and we dongt need to be killing each other here. To me that really got to me, like the message. I tried to advocate that, but the guys werengt hearing it, and they were like õfuck that. And I realized that I needed to completely cut my relationship with these guys, so I started slowly moving away.

The turning point for Emmanuel came while he was incarcerated and he had an epiphany:

Going to jail was a real eye opening moment for me. Cause with everything I did I never went to jail when I was a juvenile. That really got to me, it wasnot anything like people said, ya know? You got homies that will go out from behind bars and theyøll come out feeling like õyo man Iøm fresh out the boxö, ya know man, feeling like they have some power, ya know right? But when I went in there man, it was messing with my mind man, I was like man this is not for you. The whole time in jail, ya know you cangt be yourself. I mean you may have homies in there, but you can to be free, you miss outside, ya know? I had time to think while I was in there and I had a few people that I would call to talk to them to give me advice about what to do after I got out, cause at the time I realized most of the friends I had werengt really friends at all, ya know? And I was like man, what the heck, ya know? These guys dongt really care about me. Yeah man, I was in jail when I decided to change my life bro. I was there and I wanted to change, but I didnøt know how. So my uncle heøs actually a pastor right, so he was at a meeting and he heard about this faith-based discipleship program called [name of program] and he thought it would be good for me to change ó to change my life. So he told me if I promise to go to this program then and only then he would help bail me out.

The fact that Emmanuel was in prison and wanted to change his life yet did not know how to, highlights a gap in services and supports also noted by Okot:

They dongt have that [programs], they still dongt have it. Therege some people doing it independently trying to reach out to these kids, but no one really visits them or no one goes to jail and holds a group seminar for them and tells them of the options that they can do.

Emmanueløs experience immediately following his release from prison underscores the need for appropriate and effective programming for offenders while they

are incarcerated and highlights the pull of gang life. Emmanuel shared the following story with me regarding his release from prison:

When I came back out, two days later man I had a pocket full of crack and I had a phone. I hadnot been out of jail for like a week, ya know and we been chased like three times by the police, ya know? Once we got chased, there were three of us and we all split and I found myself just hiding behind this van, ya know and I had a clip [from a gun] in my pocket, I had money in my pocket, and I had a crack stash beside me, ya know? And I was thinking to myself oman, you remember when you were in jail and you said you were going to do anything not to come back? And now youore going to go back guaranteed! Ya know, youore going back!! On And I had all these conditions that I was breaching so I was freaked out, ya know, that I was going to back to jail. So, I just went home that night and I told my auntie who I was staying with oletos do this [name of program] thing.ö

Cassell and Weinrath's (2011) research reflects the experience of Emmanuel. In their study of justice involved gang members seeking to exit their perspective gangs, the authors found that the in-and-out process of gang-exit can take from six months to five years. Their study also found that half of their participants returned to their gangs after originally exiting them.

The push and pull of being involved in street life or gang life is evident in the stories shared by these participants. Moreover, the lure and trap of gang life also underscores the importance of sustained and targeted programming, which as the participants shared with me, was a positive influence on their lives when it was available to them. Further, Okot and Emmanuel both identified a lack of programming and available resources for young adults coming out of incarceration. Where such programming and resources do not exist at all they are obviously essential in providing incarcerated young people with options and opportunities to turn their lives around. Where such programming and resources do exist, perhaps the incarcerated young people

need to be made more aware of potential programs and opportunities that they can access and which can provide them with a constructive way out.

5.4 - Coping Strategies – "If you're ever going to get clean you have to have some way of coping." - Okot

Deciding to turn oness life around is one thing, but actually being able to sustain a healthy transition requires an ability to cope with that change. For many people the idea of physically moving away to another province or town to escape the negative influences in their lives seems like a popular idea. However, simply removing oneself geographically does not necessarily address the host of other issues, which may be negatively affecting oness behavior in the first place. As Emmanuel noted earlier, escaping negative influences is not as easy as simply moving away. As well, community worker, Saalim pointed out that parents often overreact by sending their children to live with relatives in another location only to find that the problem intensifies. For example, Okot who himself moved to Alberta for a while to escape the negative influences he was experiencing in Winnipeg shared with me the following in his story:

Thereøs like four or five groups [gangs] in Alberta, that have all left from here, ya know? They went from here to there to break away from the violence, to break away from the conflicts that were occurring. So they grew a lot more, like they have a major problem out in Alberta right now with these guys. Theyøre probably a bigger number out there now then there is here.

Bashir also moved to Alberta in search of finding a good job and cleaning up his life only to find that it wasnot what he expected:

When I moved from here I thought that I would be better too. After I got kicked out of school I wasnot doing anything here. But the job there is not even doing nothing so I need to go back to school. Iom still struggling about it, I just want to

finish my high school and get a better job. That s how you move on. Yeah, I want to finish my school first.

Moving away to escape one problems is not the only strategy for coping. Both Ermias and Okot stressed the importance of keeping themselves busy with different activities. For example, Ermias shared the following with me:

Iøm working hard man, I donøt even care about my health anymore. Iøm working everyday and Iøm trying to go back to school.

Similarly when Okot realized that it was time for him to turn his life around this is what he had to say:

I made myself really busy, ya know, full-time classes, working, seeing my son, playing basketballódoing healthy activities. I just filled my time cause I knew if I didnøt fill my time I was going to be there [in the street]. I went to a lot of AA meetings, I remember I used to go to like 5 AA meetings a week! (said with laughter) Trust me, some of these guys may want to be a kingpin, but they know what comes with it, and they donøt want to be in this game a long, if they get a chance to get outóa lot of them are like õman I just want to get out, but I donøt know how, Iøm going to make this money and thatøs it cause thatøs what I know from now until something else comes up.ö But I try and tell them, nothing is going to come up, you have to make it for yourself.

It all works hand-in-hand, ya know? If you@re ever going to get clean you have to have some way of coping. Like a lot of them cannot deal with their problems, the only way is they get high all day. Even to take one course or two courses that@s ok. If they have some kind of structure that little success that they have when they go home, they can build on that, ya know?

Like Ermias and Okot, community worker, Hassan also stated that after he dropped out of high school, he too kept himself very busy with work and activities at his local mosque in order to avoid getting himself into trouble, õI got involved in activities at the mosque, that was one of the things that saved me.ö

All of the young adult male study participants are members of ethnic communities where religion, culture, and tradition are all deeply intertwined. It is interesting therefore, that only one of these young men noted a renewed connection with his faith as the

cornerstone to helping him cope with life after leaving the gang. Emmanuel shared the journey and importance of his faith in turning his life around:

I went to [name of faith based program] with the intention of not going to jail. So I went down there with the wrong intention, but uhóthings changed. One of my mentors he told me õya know youøve been doing things by yourself the whole time and you see where it led to, ya know, and it didnøt end up good. And now, ya know if the road youøre on isnøt working in life why donøt you try something different, ya know? Why donøt you give God a chance, ya know?ö And I came to the point where I thought õyeah Iøll give God a tryö, ya know? And during that time I had a real eye opening moment, ya know, I had a different desire in me.

Further, he shared with me the importance of having positive people in his life:

The number one thing for me is accountability, ya know, having people that will hold you accountable. Real people to speak into your life to tell you like, õya know what youœre doing is wrong.ö Wise people. And staying away from things that are bad for me. Like for me the club is not my style anymore cause I used to struggle with drinking and getting high. I try to keep around me people that lift me up, that encourage me, ya know, that want the best for me, staying with the church, volunteering here and there. So for me Iøve decided that if somethingæs not working for me, if itæs not beneficial for me then Iøm not going to be there, cause that is my time.

The act of turning to faith and religion as a coping strategy in ethnopolitical conflict is an area of research which Cairns (1996) states deserves more research attention and could also be researched further within the context of gangs.

The notion of escaping a problem by moving away as three of the participants have tried, clearly does not work in and of itself, and as Okot noted has in fact spread the problem of gangs in Winnipeg into parts of Alberta. Filling the void of gang life is essential for a successful transition. The examples of Emmanuel and Okot show how filling one silfe with positive influences can be an effective way of staying away from gangs. Young people need access to programs, employment opportunities, positive mentors and activities to live a safe and productive life.

5.5 – Addressing The Issues – "We need to educate them with this knowledge of what's ahead of them." – Emmanuel

Throughout the course of their interviews the participants often shared their own thoughts about how to best address issues facing refugee young people. For example, Ermias stated that Winnipeg needs omore cops on the street, more programs for immigrants. He continued by stressing that:

The main thing is programs to keep them out of trouble so they dong get to the bad people and do bad things, ya know what I mean? Once yougre busy in the program yougre not going to care about those other thingsóyougl forget about those other people, ya know what I mean, more programs on how to get a job.

For Emmanuel the focus was on early prevention:

First, we already know what so going on and we got to recognize that and get that over with and come up with something where we can catch them early when they first come. We need to educate them with this knowledge of what sa ahead of them. What sa there, ya know? Not just wait when they are already thinking about it and seeing it and wondering, ya know? If they are trying to make that decision by themselves before you come sometimes it so late. So educate them before that and then have a plan to help them keep moving forward or to remind them just in case they are falling back or considering of going into that lifestyle, ya know, straight forward showing them the consequences. Have people like me come up and talk to them, ya know? And share stories, and show them the opportunities that they have, ya know? Sometimes we need to remind people of where they come from and if they want a better life they have to work for it. I had nothing like that.

Okot noted that it was critical to show youth the potential consequences of gang life:

You got to educate these kids about the stats, they see their friends dying, ya know, they see people going to jail, but they dongt know why that happens, ya know? If you show them the real stats like õya know, you guys by the time yougte 17 or 18 yougte either in jail or yougte dead, and if you stay in school by that time youghl be in university. All of them all they want to do is sustain their lives ya know?

Okot also stressed the importance of addressing mental health issues among refugee young people:

I think right now the main thing is like people here, like the society dong understand the amount of trauma these kids have. Like they have a lot of trauma first of all from back home. I remember one of these kids [who said] õI used to walk around with grenades packed in my bagö and he said he was 9 years old. Some people were child soldiers so that trauma plus the trauma in the street, ya know? Like you dongt understand the amount of trauma of seeing your friend get killed in the street and people who used to be your friend are your enemy now and you coming out having no safety from nobody cause everybody knows you and they know where you are, so yougre living a real messed up life, ya know? So yougre always in fearonot in fear basically, but in a õI got to protect myselfö kind of mentality and I still have to survive. On top of that the police are looking for me, I could get locked up at any time, my community hates me, my parents dongt accept me, and I cangt even go see my kids cause Igm paranoid Igm being followed, thatgs how it was for me.

So the first thing is to have a rehabilitation program for them, like therapy, they need to have group therapy, not just like therapy, group therapy because this first of all works in numbers because these kids hang out with each other, and you know they do activities togetherí They might be all five of them hanging out on the corner, but all five of them is experiencing different things, ya know? So you need to look into that, ya know, their states of trauma. Cause if they feel good about themselves, they feel good about what theyere doing, and they feel good about having a good future, or theyere hopeful, then they got a good chance at rehabilitation. The second thing is they need to have substitute options for them to have a lifestyle, thates what I had to do, find work, and had to find a life.

The need to address trauma related to violence among young refugees highlights a gap in service delivery. All of the four young male adult participants in this study have experienced physical violence in some way or another. Two of the participants shared with me they had experienced violence during war. Two had experienced violence within their homes. All four had witnessed violence in their neighborhoods in Winnipeg. All four had close friends in Winnipeg murdered as a result of gang violence. All four were victims of violence; two were admitted to hospital to treat their injuries, while three of them shared that they have also been perpetrators of violence and have spent time behind bars. Yet, none of the participants shared that they have ever been approached for or participated in any kind of therapy, which might have assisted them in coping with

whatever trauma related issues they may have. Okot noted that he has gone to counseling for his addiction issues.

Okot also noted deep structural issues that need to be addressed within the family:

This is the problem, families dongt read. They have a lot harder time cause their kids are outside theygre exposed to all these elements, but you the motherge illiterate, that father is illiterate, they dongt have an uncle here, they dongt have a Dad here, and guess what? What is happening is, ya know old dongt know what to do, I dongt know what to do, my kids are getting in trouble. If you were to teach them, if you were to teach the parents, then when the kids come home ook listen, I dongt know who your friends are, Igm not telling you not to hang out with them, but instead of hanging out with your friends all the time I want you to bring them here let them speak to me, I want to know your friendsol can bring you guys cake here, I can bring you food and I want to show you that I want to see your friends.

So educate the parents to prevent the kids from getting in trouble. Parents can also bring up issues with the kids, start using õIö statements, like õI have a problem with you guys being mad at meö, õI have trouble readingö, õI have trouble in school.ö Kids can say that, and the family can say ok õI need you to start being at home on timeö, õI need you to put up your grades.ö Then people would understand each other. When the family knows whatøs going on at home, they can take that out front and go to the community, we can recognize problems sooner, elders can make programs to address problems in the community. But they donøt do that, they just complain and say õour kids are in troubleö, but theyøre not countering that. In my opinion theyøre failing us big time!

What Okot is raising here directly correlates with much of what was shared by the participants with regards to the various frustrations and struggles the participants recognized within their families and with community elders. Further, identifying these struggles speaks to some of the gaps which exist between the child, parents, ethnic community and police as discussed earlier in this thesis.

The reality of the involvement of young refugees in gangs in Winnipeg is a multilayered issue, which requires a multi-track conflict intervention. The calls for early prevention programs for newly arrived refugees by my participants could go a long way in preparing and educating these young people about the dangers of gangs and the

potential consequences of gang involvement, as well as providing information about education and employment opportunities. As well, access to counseling and trauma healing for these young people and their families could go a long way toward healing and addressing the root causes of anti-social behavior. Further, Okotos idea of encouraging parents and their children to communicate better with one another will also serve to strengthen the bonds within families. More open communication between family members would also equip families to be better able to identify and express their concerns and worries. Confronting family issues and ensuring a stable, understanding family environment may be the foundation upon which refugee young people will also be able to thrive at school, in their neighborhoods and in the their ethnic communities.

5.6 – Key Findings of the Research

All of the participants identified positive life experiences that affected their lives. Although these experiences were not enough in and of themselves to pull my participants away from gangs, they nonetheless tell us something very important. That is, when and where programs were available, and when these young people had a positive mentor taking an interest in their lives, the young people in this study responded positively to this interaction and engagement. As many of the participants pointed to a lack of sustained programming, this positive reaction to available programming highlights the need for more available and sustained programming along-side committed mentors to walk beside such young people.

The pressure to remain loyal to the gang as noted by Okot and Emmanuel is another key finding of this research. After both of these participants had made decisions

to leave their gang they faced the threat of violence if they did not remain loyal to the gang so colors. As a result of this pressure, leaving the gang and severing ties with their offriends was a process which took time. Both Okot and Emmanuel remained connected to their gang even after they had decided to leave the gang. Emmanuel back and forth experience of trying to leave the gang speaks to the push and pull process of leaving a gang forever.

The significant life events that were ultimately the final motivating factors for my participants to turn their lives around are also key findings of this research. For Bashir and Okot seemed to have reached their capacity for the culture of violence inherent in gangs. Bashir noted for example, a violent incident in his neighborhood, which was the catalyst for him to change. Okot stated that once he was released from prison he knew he would have to kill some people in order to stay on top and he was not prepared to do that. The final push for Okot to leave his gang was the gang-related murder of one of his friends.

Where Bashir and Okotøs turning points reflected their lack of capacity to continue violent activities, Ermaisø and Emmanueløs major life changing events reflected their experience with the justice system. For Ermias, the potential to be deported was enough for him to decide to turn his life around. Similarly, Emmanueløs time spent behind bars was an awakening moment for him and it was during this time that he realized he no longer wanted to participate in gang activity.

Another key finding as noted by my participants was the perceived lack of resources and services available to them while they were incarcerated that could empower them to make positive life choices. Having various positive options and

support mechanisms available to them could assist these young men in having a successful transition back into society.

Similar to my participants who attributed much of the negative influences in their lives to the geographical neighborhood in which they were settled upon arriving in Winnipeg, so too have the parents of some of these young people also articulated this negative experience. Both the former gang member participants and the community worker participants noted how families often relocate their misbehaving youth to live with relatives in Alberta. As acknowledged in the interviews, this so called punitive act has resulted in the spread of Winnipeg based gangs such as the African Mafia into parts of Alberta. The idea of relocation seems to be short sighted and clearly does not address the root causes of why these young people are acting out in the manner in which they are behaving. Although relocation can be a good and sometimes necessary step to safely assist in transitioning someone out of a gang, without the required support mechanisms in place relocation at least as far as this study shows, does not necessarily lead to a positive turn around for young people involved in antisocial behavior and gangs.

I recently attended a conference in Toronto at York University Centre for Refugee Studies where a Somali community worker from Kitchner, Ontario shared with me that he is now seeing families move their youth from parts of Alberta to Kitchner due to the increase of violence associated with their children. Sadly, he stated that moving youth from Alberta to Kitchner has also resulted in an increase of violence in Kitchner. Thus, we are now seeing families move their youth from Winnipeg to Alberta to Kitchner all in the name of escaping gang violence. This reality should be setting off alarm bells for those concerned with the plight of refugee youth in this country as the spread of

violence from one city to the next continues. Clearly moving from one city to the next does not address the root causes of youth violence and if we as a community do not move to address issues of trauma, isolation, addiction, unemployment and poverty for example, refugee youth violence in this country will only continue to rise.

Due to the multiple stress factors contributing to their antisocial behavior, a multitrack approach is also needed where programming is concerned. Having a community sports program is good, yet if issues of family violence, trauma, unemployment, and addictions are also not addressed, it is unlikely that the sports program in and of itself will be enough to pull a young person away from antisocial behavior and gang involvement. In fact, as noted in my interviews, identifying the multiple levels of trauma associated with being a owar-affectedo refugee, and a resident of a low-income, high crime, high violence and gang ridden neighborhood is necessary in addressing the root causes of this violent and negative behavior.

5.7 - Conclusion

When these young men received programming, it proved to be a positive experience. Therefore, more resources need to be put into creating and sustaining long-term programs which not only provide these young people with recreational and employment opportunities, they should also address the root causes of their behavior. This means programs also need to include a focus on healing familial wounds, and bridging generational gaps, addressing trauma and addictions, and also training and equipping schools with the necessary resources and tools needed to meet the unique needs of refugee young people living in Winnipeg

Chapter 6 – Finding a Way Forward

6.1 - Introduction

It is clear from the findings of this study that refugee young people face numerous challenges on a daily basis in areas such as: school, neighborhood, peer group, family, ethnic community and law enforcement. Each of these areas comes with its own set of unique issues, which need to be addressed in an appropriate fashion.

6.2 – Reflecting on Key Findings

With regards to challenges in school such as inappropriate grade placement and language barriers, it is clear that at the time my participants were attending school, the schools were not equipped to address their unique needs and challenges. As noted in the literature review, teachers should be equipped with the appropriate training and resources needed to more effectively meet the needs of refugee youth, and to better understand potential anti-social behavioral patterns (Kanu, 2008, 2009; Rossiter & Rossiter, 2009; Sersli et al., 2010; Byrne & Senehi, 2012). However, it is important to note that efforts are currently being made to address these gaps. The work of Jan Stewart (2011) is important in this regard, as listening to the voices of refugee young people who are in the Winnipeg school system is essential for developing the appropriate actions needed to address their unique issues, challenges and concerns within their Winnipeg school experience.

The notion of discipline within schools with regards to the expulsion of all four young adult male participants is a cause for concern. In all four cases it was after they were expelled from school that they really started into a downward spiral heading toward

entrenched gang life. It is obvious that schools have rules and regulations and there needs to be consequences for one actions, however expelling a student does not address the root causes of why that student was acting out in the first place. Rossiter and Rossiter (2009) and Sersli, Salazar and Lozano (2010) state that youth may act out in violence as a way to cope with stress. Moreover, Noguera (2000) argues that youth may act out in violence if they are osocially ostracized, bullied, teased, or mistreated by school officials or other studentso (p. 135). Indeed, as evidenced in my findings, my participants largely felt ostracized, bullied, and not cared for within their schools. MacLeod (2009) argues that more attention needs to be brought to bear upon the school and how it places value on olower-classo children. He states of the problem is not that lower-class children are inferior in some way; the problem is that by the definitions and standards of the school, they consistently are evaluated as deficient (p. 101).

The policy of expulsion sends an already troubled youth into the waiting arms of gangs. Once a youth is expelled from a school it becomes very difficult for them to be accepted into another school. As my results indicate, many never go back to school. These findings corroborate the findings of Polakow-Suransky (2000) who also found that most young people do not return to school after being expelled, and in fact often continue down a negative path. Of course school administrators and educators need to ensure the wellbeing of the school population on a whole, but if we truly value our young people we must try to address the root causes of these actions and provide the necessary support systems to keep these young people on a positive path.

As noted by Okot, problems at school can often be a result of problems at home.

My young male participants all struggled with internal family conflict. According to

Kanu (2008, 2009), Rossiter and Rossiter (2009), Sersli et al. (2010), Stewart (2011), and Byrne and Senehi (2012) this is not a surprise as refugee single mothers struggle to make ends meet and are often not able to assist their children with school related work due to their busy schedule and often inadequate education. Further, Rossiter and Rossiter (2009), Sersli et al. (2010), and Shakya et al. (2010) note that trauma experienced by parents (for example, Okotøs father) prevents them from being a stabilizing force in the family.

In order to combat the lack of parental and community support for their children Okot called for parents, children, and the broader ethnic community to communicate with one another and to listen to each other¢s needs. Indeed, such family support mechanisms have proven to be effective in steering young people away from violence and other antisocial behavior (Byrne, McLeod & Polkinghorn, 2004; Wortley & Tanner, 2007; Molix & Bettencourt, 2010; Sersli et al., 2010).

The negative influences within one neighborhood and within one peer group can also lead to conflicts at home and stigmatization within one ethnic community as evidenced by several of the study participants. Chekki (2006), Rossiter and Rossiter (2009), and Sersli et al. (2010) note that ethnic communities can play a key role in ensuring the resiliency of refugee youth by providing them with spiritual guidance, cultural identity, and a sense of belonging. However, according to the perceptions and experiences of the young male refugees and community worker participants in this study, the ethnic communities of these young men failed to meet their needs. Rather than providing a system of support for these young men, the leaders in their ethnic communities were perceived as being judgmental and dismissive. Okot and the other

participants called for more action on the part of community leaders as well as more targeted programming that informs these young people of the potential dangers in their communities and the consequences of engaging in gang activity. The views of my participants correlate with those of Chekki (2006), Sersli et al. (2010), and Byrne and Senehi (2012) who all stress the importance and effectiveness of community leaders, mentors and elders in guiding young people down a positive constructive path.

The negative experiences of my participants in terms of school, family, ethnic community, neighborhood, peer group, as well as with law enforcement, should not be addressed in isolation from each other. The cyclical and interwoven nature of these issues requires a holistic and multilayered conflict intervention approach. If a young person does not have a solid home environment with engaged parents, it is unlikely that they will be constructively engaged in school, which means it is more likely they will be engaged with negative peers and anti-social behavior in their neighborhoods. As evidenced through the experiences of my young male participants, anti-social behavior and gang activity leads to negative experiences with law enforcement, which leads to stigmatization from their ethnic community and further conflict in the home, and the cycle continues to repeat itself. Nanette J. Davis (1999) asserts that õa widely recognized gang characteristic includes alienation from the traditional foundations of community ó home, school, work, as well as adult and elderly role models and associates ó which serves to isolate the gang, rendering it apart from the fabric of societyö (p. 231).

Calls for parents and community members to become more engaged in the lives of their young people are too simplistic if they do not consider the unique challenges that refugee parents and community elders themselves have. As evidenced through my

interviews, refugee single mothers are often busy working trying to make ends meet and do not have time for extracurricular meetings. Fathers too face challenges of lowered social status. Further, community leaders and families in general are busy simply trying to build a new life for themselves. Suarez-Orozco and Suarez-Orozco (2001) note that õmigration tends to have a destabilizing effect on the familyö (p. 75) due to the business of building a new life, which renders parents less available for their children often leaving them õto their own devices long before they are developmentally readyö (p. 75).

Based on my study evidence it is true that refugee parents and community elders need to become more involved in the lives of their youth. The literature also stresses the importance of community noting that:

Community is a place where adolescents feel a sense of belonging and connections with others in a special way. It is a place where they learn about who they are. It is where they begin to develop a sense of identity and a vision of how the fit in the world around them. Community is a place where adolescents can find answers to life many difficult and complicated questions (Hardy & Laszloffy, 2005, p. 63).

Appropriate programming and education for ethnic communities and parents about how to understand the issues their youth are facing were mentioned by the studyøs participants as intervention processes that are needed. This study has shown that current disciplinary measures such as relocating a troubled youth to live with relatives in Alberta simply are not working. Rather, such measures have in fact made the issue worse by geographically expanding the õtroubleö into other parts of Canada. However, a significant challenge to organizing such a strategy as evidenced in the findings is the difficulty parents and their children have finding the time to attend such programs. There is no easy answer to this issue, and the solution requires further discussion with the parents and community elders

themselves in order to identify their particular needs that would empower them to be able to attend such programs.

Collaborative efforts such as *Umoja* which aims to bring together parents, elders, community agencies, and members of the Winnipeg Police Service in order to build mutual understanding and trust are definitely a step in the right direction. However, at times it has been difficult getting members of the community to attend events, which again requires further discussion with parents and community elders in order to find a solution that will encourage them to attend these events. Collaborative efforts such as *Umoja* must include the voices of young people as well as educators in order to provide a well-rounded and informed discussion on issues affecting refugee communities.

The participants of this study faced many negative experiences and challenges in school, family, neighborhood, ethnic community, peer group, and with law enforcement. The lack of support systems in these six areas resulted in the unfulfilled basic needs of my study participants. These universal needs cannot be negotiated (Burton, 1993; Jeong, 2000; Byrne & Senehi, 2012) and as evidenced in my research findings, and as noted by Sersli et al. (2010), where traditional support systems have failed to meet the basic needs of young people, gangs are more than willing to fill the void.

Although they did not name it explicitly, the young men in this study recognized that they faced various levels of structural violence or social injustice in their lives. The inequalities in power relations, opportunities and distribution and access to resources that define structural violence (Galtung, 1969; Byrne & Senehi, 2012) were experienced by my participants in a variety of ways. First, the young men in this study were acutely aware that their families were settled in a particular neighborhood that was infested with

crime, drugs and gangs and they had no power to say otherwise. All of the young men in this study thought that if they would have had a choice to be settled in another neighborhood such as St. Vital or Tuxedo they would not have fallen into gang life.

Second, the young men in this study were placed in school classes based on their age rather than on their level of English comprehension. The policy of placing a young refugee in a class based on her or his age when in fact s/he may have had little or no prior education sets them up to fail especially where schools are not equipped with the necessary resources to help these young people succeed.

Third, both the young men and community workers felt that refugee young people were unfairly targeted and treated by law enforcement officers in Winnipeg. The racism and harassment experienced by my study participants from members of the Winnipeg Police Service in some ways echoes the experiences of many refugees with systems of authority in their countries of origin where many have faced racism and other forms of exclusion and harassment (Stewart, 2011; Suarez-Orozco & Suarez-Orozco, 2001; Wolf, 2006). Further, these negative experiences with law enforcement officers serve to reinforce negative perceptions of authority some refugee youth already have due to past experiences in their country of origin.

The literature on structural violence and basic human needs asserts that physical violence can arise as people respond to the injustice of being prevented from achieving their full potential and in fulfilling their basic needs (Galtung, 1969; Staub, 1989; Burton, 1993; Gilligan, 1996; Jeong, 2000; Byrne & Senehi, 2012). Indeed, the young male participants in this study experienced physical violence as witnesses, victims, and

perpetrators in their homes, at school, and in their neighborhoods as a result of unfulfilled basic needs and structural violence.

The use of physical violence by my participants as a way to cope with challenges at home and at school reflect the assertions of Rossiter and Rossiter (2009), and Sersli et al. (2010) who contend that youth may use physical violence to cope with stressors in their lives. Ermias for example, saw physical violence as a natural response to the violence already occurring in his neighborhood. This matter of fact approach to physical violence reflects the notion of Kanu (2008) and Byrne and Senehi (2012) who contend that young people become desensitized to physical violence if they are continually exposed it in their daily lives.

The positive experiences noted by my study participants, highlight the powerful impact engaged and passionate educators can have on young people. Further, when the participants had access to programming whether it was, music, sports, employment, spiritual mentoring or counseling, all of them stated that these programs affected their life in a positive way. Vorrasi and Garbarino (2000) highlight the importance of programming by asserting that õrisk factors may be neutralized or at least partially offset by the introduction of opportunity factors into other realms of the child life, even when patterns of risk are thought to be impervious to interventionö (p. 63). Not surprisingly, all of the participants expressed the need for more programming geared towards refugee youth in the areas of trauma counseling, sports, music, skills training, and employment opportunities.

Much of the funding and resources needed for such programs is often provided by the various levels of government. Thus, where governments do not have the will, desire, or resources to fund such programming, service providers must look elsewhere. Alternatively, there must be a collaborative effort on the part of young people, parents, community elders, community organizations, businesses, educators and law enforcement to inform government of these gaps in service provision and how ensuring these services will benefit society as a whole.

The results of this study have shown that the challenges faced by refugee young people in Winnipeg are complex and multilayered. The causes of my young male participantsø anti-social behavior and gang involvement were also multilayered. This finding corroborates with the work of Howard and Jensson (1999), Vorrasi and Garbarino (2000), and Garbarino (2001) who argue that youth violence is the result of young people living with multiple risk factors. Thus, Byrne and Senehi (2012) contend that conflict resolution intervention approaches to addressing these issues need to be multilayered as well. By looking at the õweb of interconnected partsö, which function together, Diamond and McDonald (1996) contend that we are better able to õsee who we are, what weare doing, how weare doing it, where weare headed, and whyö (p. 1). Law enforcement, early prevention programs, educational programs, employment programs, sports and music programs, community engagement, and family and individualized therapy programs all need to be part of the solution. Where such services exist young people need to be made aware of them.

The importance of being made aware of existing social services is especially important for those young people wanting to leave their gang. As evidenced in my research findings, many gang-involved young people do in fact want to exit their gangs, yet they simply do not know where to turn for help. The in-and-out process of leaving

the gang and no longer associating with people one considered family as shared by both Emmanuel and Okot underscores, the difficulty of gang-exit and highlights the importance of local community resources to empower young people to make the right choices during such vulnerable and critical times. The findings of Cassell and Weinrath (2011) who note the lengthy in-and-out process of gang-exit as well as the high rate of return of ex-gang members to their gangs, further stresses the importance of targeted programming during the critical in-and-out gang-exit process. Their findings also reflect the sentiments of Emmanuel and Okot who noted the need for support services for young people who are coming out of correctional institutions while they are transitioning back into society.

In order to achieve the goal of increasing services, there needs to be a concerted effort on the part of community leaders, educators, religious leaders, law enforcement, families, community agencies, corrections, probation services, government agencies, and the business community to collaborate and work together towards these goals. As Diamond and McDonald (1996) suggest, a multi-track peacebuilding effort requires us to acknowledge our interrelatedness and work together towards a shared goal rather than in isolation of each other. We must seek out and listen to the voices of these young people who have lived these experiences, who have been in gangs, who have been locked up, and who feel abandoned by their community. It is the voices of young people in particular that must inform policy and programming as we move forward.

6.3 - Future Research

Future research with regards to gang-affected young refugees living in Winnipeg should consider examining the programs and resources, which already exist that aim to assist gang involved young people transition into a more positive lifestyle. More research also needs to be carried out on which refugee communities are most affected by gangs, and which refugee communities, if any, are unaffected. Understanding these differences about why African youth are joining gangs, compared to refugee youth from Myanmar or Iraq, for example, will help us identify possible gaps in the available support systems. As well, understanding whether French speaking African communities are as affected by gangs as non-French speaking African communities may help to inform gang prevention/intervention programming and guide the distribution of resources appropriately. These differences will assist in identifying points of resilience and support structures that may exist in some communities but not in others which can be used as models to be adapted by communities who are facing gang involvement issues amongst their youth.

The original proposal for this study included the participation of female formerly gang involved or gang associated participants. Unfortunately, as noted earlier I did not get to interview any female former gang members. Although most of the academic literature centered on gangs focuses on males, females also play a key part in gang life. It is clear from speaking with my participants that female refugee youth are also involved in gangs in Winnipeg. However, it appears that there is no programming geared specifically for female gang involved refugee youth, and that they are hard to identify because they do not have as much involvement with law enforcement as their male counterparts.

More work also needs to be done in connecting with parents to determine their unique needs and challenges. It is clear that a gap exists between refugee young people and their parents. Future research needs to explore what resources are available to parents, which may assist them in addressing the unique and unfamiliar challenges in parenting which this new context of living in Winnipeg has brought them. In particular, what are the resources available to parents informing them of what to look for regarding the potential gang involvement of their child, or what are the specific signs of drug addiction? What are the resources available to them to assist in parenting a child who is going down the path to gang involvement? Even if these specific resources do exist, many parents may not know how to access them.

Certainly more research needs to be carried out on the relationship between refugee young people and their specific ethnic communities. Although the relationship between refugee young people and their ethnic communities has been identified as a big gap, some positive steps are being taken within the South Sudanese Dinka community to bridge the generational gap. As recently as May 4th, 2013 the community hosted a fundraising event led by its youth which raised money to support Dinka cultural programs for the young people, as well as supporting programs geared toward fostering parent-youth dialogue. This initiative could go a long way in bridging the gap between generations. It will be interesting to see how this initiative measures success and how this new intervention process will affect the behavior and identity of Dinka youth and the relationships they have with their parents and their elders in the community.

Finally, as refugee based gangs from Winnipeg spread into Alberta and across the country, research needs to be carried out on a Canada wide basis. Examining how other

cities are addressing refugee youth gang issues can be the beginning of building cross-Canada collaborative efforts to provide the necessary support mechanisms and programming for these young people no matter which province they might happen to be in.

6.4 - Final Conclusion

This qualitative research study examined the perceptions, challenges and experiences of young adult male refugees living in Winnipeg. I sought to gain a better understanding of how and why these young people developed negative behavior patterns and consequently became involved in street gangs. Further, I have sought to understand what has motivated them to turn their lives around for the better. It is hoped that the data collected during this study will be useful in informing various service providers, policy makers, and peace workers about ongoing challenges and possible gaps in service provision for refugee young people living in the city of Winnipeg.

It is clear that where schools, families, and ethnic communities have failed to meet the needs of these young people, negative support systems such as gangs found within their neighborhoods are ready to fill the void. In a time of war youth may join an armed struggle to fulfill their needs for protection, belonging, family, justice, food, clothing and shelter (Wessells, 2006; Wessells & Jonah, 2006). Similarly, in this study young people have joined a gang to fulfill those same needs. Wessells (2006) states that where support systems such as family, teachers, and community leaders exist, youth living in war zones show a higher rate of resiliency. As refugee youth are now living in the urban owar zones of our cities where drugs, crime, gangs and violence permeate

through the neighborhoods in which they live, we need to ensure that these same supports systems exist to provide for their needs.

References

- Arendt, H. (1970). On violence. San Diego: Harcourt Brace & Company.
- Barriball, K. L., & While, A. (1994). Collecting data using a semi-structured interview: a discussion paper. *Journal of Advanced Nursing*, *19*(2), 328-35. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/8188965
- Batiskunda, R. (2009). The Rwandan conflict. In J. Carter, G. Irani, & V. D. Volkan (Eds.), *Regional and ethnic conflicts: Perspectives from the front lines* (pp. 133-158). Upper Saddle River: Prentice Hall.
- Beah, I. (2007). A long way gone. Vancouver: Douglas & McIntyre.
- Bemak, F., & Keys, S. (2000). Violent and aggressive youth: Intervention and Prevention strategies for changing times. Thousand Oaks: Corwin Press, INC.
- Bogdan, R. C., & Knopp Biklen, S. (2003). *Qualitative research for education: An introduction to theories and methods* (fourth.). Boston: Pearson Education Group Inc.
- Brodbeck, T. (2013, April). Less downtown crime overall? That & BS. Winnipeg Sun. Winnipeg. Retrieved from http://www.winnipegsun.com/2013/04/04/less-downtown-crime-overall-thats-bs
- Bryman, A., & Bell, E. (2007). Interviewing in qualitative research. *Business research methods* (2nd ed., pp. 471-508). Oxford: Oxford University Press. Retrieved from http://books.google.ca/books?id=vpqjq4I9KGMC&pg=PA471&lpg=PA471&dq=õi nterviewing+in+qualitative+researchö+õthis+chapter+is+concerned+with+the+inter view+in+qualitative&source=bl&ots=_dxfrE9rzR&sig=sV7Lfqe49SN5ECa_fJE2Z MayJVI&hl=en&sa=X&ei=dAVuT9vADcbf0QG7mfDRBg&ved=0CCYQ6AEwA Q#v=onepage&q=öinterviewing in qualitative researchõ öthis chapter is concerned with the interview in qualitative&f=false
- Burton, J. W. (1993). Conflict resolution as a political philosophy. In D. J. D. Sandole & H. van der Merwe (Eds.), *Conflict resolution theory and practice: Integration and application* (pp. 55-64). Manchester: Manchester University Press.
- Burton, J. W. (Ed.). (1990). Conflict: Human needs theory. New York: St. Martings Press.
- Byrne, S. (1997). Belfast schoolchildrenøs images of political conflict and social change: Signs of hope in integrated education. *Mind and Human Interaction*, 8(3), 172-185.
- Byrne, S., & Senehi, J. (2012). *Violence: Analysis, intervention, and prevention*. Athens: Ohio University Press.

- Byrne, S., McLeod, C., & Polkinghorn, B. (2004). University students from four ethnopolitical conflict zones: An exploration study of perceptions of self and country. *Peace and Conflict Studies*, *11*(2), 13-34. Retrieved from http://shss.nova.edu/pcs/journalsPDF/V11N2_Abstracts.pdf
- Cairns, E. (1996). Children and political violence. Oxford: Blackwell Publishers.
- Carter, T. (2009a). *Planning for Newcomers in Winnipeg's Inner City. Statistics* (pp. 1-7). Retrieved from www.http://canada.metropolis.net/pdfs/Planning_for_Newcomers_e.pdf
- Carter, T. (2009b). *The resettlement experience of privately sponsored refugees in Winnipeg*, *Manitoba*. *Statistics* (pp. 104-108). Retrieved from http://canada.metropolis.net/publications/odc09_pdfs/TomCarter_ODC09.pdf
- Carter, T., Polevychok, M., & Friesen, A. (2006). *Winnipeg's inner-city: Research on the challenges of growing diversity* (pp. 50-55). Retrieved from http://canada.metropolis.net/publications/diversity/our_diverse_cities_vol2_en.pdf
- Cassell, C., & Weinrath, M. (2011). Barriers to Leaving the Gang: An Exploratory Analysis. *The Annual Review of Interdisciplinary Justice Research*, 2, 72-99.
- Chekki, D. A. (2006). Immigrant and refugee serving organizations in a Canadian city: An exploratory study. Retrieved from http://pcerii.metropolis.net/WorkingPapers/WP01-06.pdf
- Coles, R. (1986). *The political life of children* (1st ed.). Boston: The Atlantic Monthly Press.
- Davis, N. J. (1999). *Youth crisis: Growing up in the high-risk society*. Westport: Praeger Publishers.
- DiCicco-Bloom, B., & Crabtree, B. F. (2006). The qualitative research interview. *Medical Education*, 40(4), 314-21. doi:10.1111/j.1365-2929.2006.02418.x
- Diamond, L., & McDonald, J. (1996). *Multi-track diplomacy: A systems approach to peace* (3rd ed.). West Hartford: Kumarian Press.
- Druckman, D. (2005). *Doing research: Methods of inquiry for conflict analysis*. Thousand Oaks: Sage Publications.
- Fisher, R. J. (1990). Needs theory, social identity and an eclectic model of conflict. In J. Burton (Ed.), *Conflict: Human needs theory* (pp. 89-114). New York: St. Martinøs Press.

- Fitzgerald, R., Wisener, M., & Savoie, J. (2004). *Neighbourhood characteristics and the distribution of crime in Winnipeg. Statistics* (pp. 1-63). Ottawa. Retrieved from http://www.publications.gc.ca/site/archivearchivearchived.html?url=http://www.publications.gc.ca/collections/Collection/Statcan/85-561-MIE/85-561-MIE2004004.pdf
- Galtung, J. (1969). Violence, peace, and peace research. *Journal of Peace Research*, 6, 167-191. doi:10.1177/002234336900600301
- Garbarino, J, Kostelny, K., & Dubrow, N. (1991). What children can tell us about living in danger. *The American psychologist*, *46*(4), 376-83. Retrieved from http://www.ncbi.nlm.nih.gov/pubmed/2048796
- Garbarino, J. (2001). Lost boys: Why our sons turn violent and how we can save them. *Smith College Studies in Social work*, 71(2), 167-181. doi:http://dx.doi.org/10.1080/00377310109517622
- Garbarino, J., Kostelny, K., & Dubrow, N. (1991). *No place to be a child: Growing up in a war zone*. Lexington: D.C. Heath and Company.
- Gilligan, J. (1996). *Violence: Reflections on a national epidemic*. New York: Random House.
- Grossman, D. (1996). *On killing: The psychological cost of learning to kill in war and society.* Boston: Little, Brown and Company.
- Hardy, K. V., & Laszloffy, T. A. (2005). *Teens who hurt: Clinical interventions to break the cycle of adolescent violence*. New York: The Guilford Press.
- Helsing, J., Kirlic, N., McMaster, N., & Sonnenschein, N. (2006). Young people's activism and the transition to peace. In S. McEvoy-Levy (Ed.), *Troublemakers or peacemakers? Youth and post-accord peace building* (pp. 195-216). Notre Dame: University of Notre Dame Press.
- Howard, M. O., & Jenson, J. M. (1999). Causes of youth violence. In J. M. Jenson & M.O. Howard (Eds.), *Youth violence: Current research and recent practice innovations*. Washington: NASW Press.
- Hyder, T. (2005). War, conflict and play. Berkshire: Open University Press.
- Jeong, H.-W. (2000). *Peace and conflict studies: An introduction*. Burlington: Ashgate Publishing Company.
- Johnston, L. M. (2005). Narrative analysis. In D. Druckman (Ed.), *Doing research: Methods of inquiry for conflict analysis.* Thousand Oaks: Sage Publications.

- Kanu, Y. (2008). Educational needs and barriers for African refugee students in Manitoba, *Canadian Journal of Education 31*(4), 915-940. Retrieved from http://www.csse-scee.ca/CJE/Articles/FullText/CJE31-4/CJE31-4-Kanu.pdf
- Kanu, Y. (2009). Educational pathways to successful social integration for African refugee students in Manitoba (pp. 114-119). Retrieved from http://canada.metropolis.net/publications/odc09_pdfs/YattaKanu_ODC09.pdf
- Kaplan, I. (2009). Effects of trauma and the refugee experience on psychological assessment processes and interpretation. *Australian Psychologist*, 44(1), 6-15. doi:10.1080/00050060802575715
- Kirbyson, G. (2012, September 11). Helping Central Park reach its full potential. *Winnipeg Free Press*. Winnipeg. Retrieved from http://www.winnipegfreepress.com/local/helping-central-park-reach-its-full-potential-178052381.html
- Lederach, J. P. (2005). *The moral imagination: The art and soul of building peace*. oxford: Oxford University Press.
- Luborsky, M. R., & Rubinstein, R. L. (1995). Sampling in qualitative research: Rationale, issues, and methods. *Research on aging*, *17*(1), 89-113. doi:10.1177/0164027595171005
- MacLeod, J. (2009). Ain't no makin' it: Aspirations and attainment in a low-income neighborhood (3rd ed.). Boulder: Westview Press.
- Macdonald, N. (2007, August). From Africa, with guns. *Macleans*. Retrieved from http://www.macleans.ca/canada/features/article.jsp?content=20070827_108301_108 301
- McCluskey, K. W., & Torrance, E. P. (2003). *Mentoring: One pathway to Aboriginal talent development*. Winnipeg. Retrieved from http://ius.uwinnipeg.ca/WIRA/PDF/mentoring_aboriginal_devel_interim.pdf
- McCutcheon, R. (2004). Global processes and violence: Reflections on war with Iraq. In J.-L. Chodkiewicz & R. E. Wiest (Eds.), *Globalization and community: Canadian perspectives* (pp. 179-190). Winnipeg: University of Manitoba Anthropology Papers 34.
- Miranda, D., & Claes, M. (2004). Rap music genres and deviant behaviors in French-Canadian adolescents. *Journal of Youth and Adolescence*, 33(2), 113-122. doi:10.1023/B:JOYO.0000013423.34021.45

- Molix, L., & Bettencourt, B. A. (2010). Predicting well-being among ethnic minorities: Psychological empowerment and group identity. *Journal of Applied Social Psychology*, 40(3), 513-533. doi:10.1111/j.1559-1816.2010.00585.x
- Neufeld-Redekop, V. (2002). From violence to blessing: How an understanding of deeprooted conflict can open paths to reconciliation. Ottawa: Novalis.
- Neumann, M. S. (2011). Peace promotion among ethnically diverse youth: Reflection on an agency's vision. Trinity Western University.
- Noguera, P. A. (2000). How student perspectives on violence can be used to create safer schools. In V. Polakow (Ed.), *The public assault on America's children: Poverty, violence, and juvenile injustice* (pp. 130-153). New York: Teachers College Press.
- Nordstrom, C. (1997). *A different kind of war story*. Philadelphia: University of Philadelphia Press.
- Oppenheimer, L., Bar-Tal, D., & Raviv, A. (1999). Introduction: Understanding peace, conflict, and war. In A. Raviv, L. Oppenheimer, & D. Bar-Tal (Eds.), *How children understand war and peace* (pp. 1-20). San Fransisco: Jossey-Bass Publishers.
- Polakow-Suransky, S. (2000). America least wanted: Zero-tolerance policies and the fate of expelled students. In V. Polakow (Ed.), *The public assault on America's children: Poverty, violence, and juvenile injustice* (pp. 101-129). New York: Teachers College Press.
- Polkinghorn, B., & Byrne, S. (2001). Between war and peace: An examination of conflict management styles in four conflict zones. *International Journal of Conflict Management*, 12(1), 23-46.
- Prothrow-Stith, D., & Spivak, H. R. (2004). *Murder is no accident*. San Francisco: Jossey-Bass Publishers.
- Rossiter, M. J., & Rossiter, K. R. (2009). *PMC working paper series immigrant youth and crime: Stakeholder perspectives on risk and protective factors*. Retrieved from http://pcerii.metropolis.net/WorkingPapers/WP0209.pdf
- Roy, R. (1990). Social conflicts and needs theories: Some observations. In J. Burton (Ed.), *Conflict: Human needs theory* (pp. 125-148). New York: St. Martinøs Press.
- Rubin, A., & Babbie, E. R. (2011). *Research methods for social work* (7th ed.). Belmont: Cengage Learning.
- Senehi, J. (2009). Building peace: Storytelling to transform conflicts constructively. In D. J. Sandole, S. Byrne, I. Sandole-Staroste, & J. Senehi (Eds.), *Handbook of conflict analysis and resolution*. London: Routledge.

- Senehi, J., & Byrne, S. (2006). From violence toward peace: The role of storytelling for youth healing and political empowerment after social conflict. In S. McEvoy-Levy (Ed.), *Troublemakers or peacemakers? Youth and post-accord peace building* (pp. 235-258). Notre Dame: University of Notre Dame Press.
- Sersli, S., Salazar, J., & Lozano, N. (2010). *Gang prevention for new immigrant and refugee youth in B*. *C*. *community consultation paper*. *Society* (pp. 1-33). Retrieved from www.pssg.gov.bc.ca/crimeprevention/publications/docs/gang-prevention-immigrant-refugee.pdf
- Shakya, Y. B., Khanlou, N., & Gonsalves, T. (2010). *Determinants of mental health for newcomer youth: Policy and service implications*. Retrieved from http://canada.metropolis.net/pdfs/immi_health/Immigrant Mental Health pgs98-102.pdf
- Smith, L. T. (1999). *Decolonizing methodologies: Research and Indigenous peoples*. London: Zed Books Ltd.
- Staub, E. (1989). *The roots of evil: The origins of genocide and other group violence*. New York: Cambridge University Press.
- Stewart, J. (2011). Supporting refugee children: Strategies for educators. Toronto: University of Toronto Press.
- Suarez-Orozco, C., & Suarez-Orozco, M. M. (2001). *Children of immigration*. Cambridge: Harvard University Press.
- The crimestat website. (2013). Retrieved from http://www.winnipeg.ca/crimestat/
- Turner, J. (2013). Two arrests in fatal shooting at Johnny Gøs restaurant on Main Street. Retrieved from http://www.winnipegsun.com/2013/03/07/two-arrests-in-fatal-shooting-at-johnny-gs-restaurant-on-main-street
- UNHCR. (2010a). *UNHCR Statistical Yearbook 2010 Population levels and trends* (pp. 20-35). Retrieved from www.unhcr.org/4ef9c7849.html
- UNHCR. (2010b). *UNHCR Statistical Yearbook 2010 Demographic characteristics and type of location* (pp. 47-51). Retrieved from http://www.unhcr.org/4ef9c83f9.html
- *United Nations 65th General Assembly report.* (2010). Retrieved from www.un.org/News/Press/docs/2010/gashc3992.doc.htm
- Vorrasi, J. A., & Garbarino, J. (2000). Poverty and youth violence: Not all risk factors are created equal. In V. Polakow (Ed.), *The public assault on America's children: Poverty, violence, and juvenile injustice* (pp. 59-77). New York: Teachers College Press.

- Wessells, M. (2006). Child soldiers. Cambridge: Harvard University Press.
- Wessells, M., & Jonah, D. (2006). Recruitment and reintegration of former youth soldiers in Sierra Leone: Challenges of reconciliation and post-accord peace building. In S. McEvoy-Levy (Ed.), *Troublemakers or peacemakers?: Youth and post-accord peace building* (pp. 27-47). Notre Dame: University of Notre Dame Press.
- Wilkinson, L. (2002). Factors influencing the academic success of refugee youth in Canada. *Journal of Youth Studies*, *5*(2), 173-193.
- Wilson, S. (2008). *Research is ceremony: Indigenous research methods*. Halifax: Fernwood Publishing.
- Wolff, S. (2006). Ethnic conflict: A global perspective. Oxford: Oxford University Press.
- Wortley, S., & Tanner, J. (2007). Criminal organizations or social groups? Retrieved from http://canada.metropolis.net/pdfs/WortleyTanner2007.pdf

Appendix A Interview Guide

| Question #1: |
|--|
| How did you come to Winnipeg? |
| |
| Question #2 |
| What are some of the challenges you have faced as a refugee? |
| |
| Question #3 |
| What are some of the coping strategies you use to get you through hard times? |
| |
| Question #4 |
| What is one experience that sticks out in your mind as a newcomer to Winnipeg? |
| |
| Question #5 |
| What were some of the key turning points in your life? |
| |
| Question #6 |
| What are your worst fears and worries for your life in the future? |
| |
| Question #7 |
| What are your best hopes and dreams for your life in the future? |